



# **A Comprehensive Review on Techno-Economic Analysis and Optimal Sizing of Hybrid Renewable Energy Sources with Energy Storage Systems**

Takele Ferede Agajie <sup>1,2</sup>, Ahmed Ali <sup>3,\*</sup>, Armand Fopah-Lele <sup>4</sup>, Isaac Amoussou <sup>1</sup>, Baseem Khan <sup>5,6</sup>, Carmen Lilí Rodríguez Velasco <sup>6,7</sup> and Emmanuel Tanyi <sup>1</sup>

- <sup>1</sup> Department of Electrical and Electronic Engineering, Faculty of Engineering and Technology, University of Buea, Buea P.O. Box 63, Cameroon
- <sup>2</sup> Department of Electrical and Computer Engineering, Debre Markos University, Debre Markos P.O. Box 05, Ethiopia
   <sup>3</sup> Department of Electrical and Electronic Engineering Technology, Faculty of Engineering and the Built Environment,
- Department of Electrical and Electronic Engineering Technology, Faculty of Engineer University of Johannesburg, Johannesburg 2092, South Africa
   Department of Machanical Engineering Faculty of Engineering and Technology
- <sup>4</sup> Department of Mechanical Engineering, Faculty of Engineering and Technology, University of Buea, Buea P.O. Box 63, Cameroon
- <sup>5</sup> Department of Electrical and Computer Engineering, Hawassa University, Hawassa P.O. Box 05, Ethiopia
- <sup>6</sup> Department of Project Management, Universidad Internacional Iberoamericana, Campeche 24560, Mexico
- <sup>7</sup> Department of Project Management, Universidad Europea del Atlántico, C/Isabel Torres 21, 39011 Santander, Spain
- Correspondence: aali@uj.ac.za

Abstract: Renewable energy solutions are appropriate for on-grid and off-grid applications, acting as a supporter for the utility network or rural locations without the need to develop or extend costly and difficult grid infrastructure. As a result, hybrid renewable energy sources have become a popular option for grid-connected or standalone systems. This paper examines hybrid renewable energy power production systems with a focus on energy sustainability, reliability due to irregularities, techno-economic feasibility, and being environmentally friendly. In attaining a reliable, clean, and cost-effective system, sizing optimal hybrid renewable energy sources (HRES) is a crucial challenge. The presenters went further to outline the best sizing approach that can be used in HRES, taking into consideration the key components, parameters, methods, and data. Moreover, the goal functions, constraints from design, system components, optimization software tools, and meta-heuristic algorithm methodologies were highlighted for the available studies in this timely synopsis of the state of the art. Additionally, current issues resulting from scaling HRES were also identified and discussed. The latest trends and advances in planning problems were thoroughly addressed. Finally, this paper provides suggestions for further research into the appropriate component sizing in HRES.

**Keywords:** hybrid energy system; reliability analysis; techno-economic analysis; optimization methods; energy storage option; energy management system

# 1. Introduction

The electrical loads in residential, commercial, local, and industrial buildings have increased dramatically because of the increased reliability of fossil fuels for energy. Use of renewable energy sources for electric power generation supply should be prioritized in order to reduce electric load dependency on fossil fuels. Currently, crude oil, coal, and natural gas are used as alternative energy sources to meet about 70% of the global power demand [1]. Energy demand is skyrocketing in response to the world's growing economy and population. Consequently, fossil fuel consumption is also increasing very steeply. Conventional fuel stocks are limited and rapidly declining, which requires immediate action and long-term solutions to avoid a possible energy disaster in the years to come. Furthermore, fossil fuels are potential sources of hazardous emissions, such as greenhouse gases, which greatly contribute to warming the globe [2,3].



Citation: Agajie, T.F.; Ali, A.; Fopah-Lele, A.; Amoussou, I.; Khan, B.; Velasco, C.L.R.; Tanyi, E. A Comprehensive Review on Techno-Economic Analysis and Optimal Sizing of Hybrid Renewable Energy Sources with Energy Storage Systems. *Energies* 2023, *16*, 642. https://doi.org/10.3390/ en16020642

Academic Editors: Charles Egbu, Tariq Umar and Nnedinma Umeokafor

Received: 5 December 2022 Revised: 1 January 2023 Accepted: 3 January 2023 Published: 5 January 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

To be able to reduce global warming, one of the techniques is to raise awareness of the significance of reducing power usage in homes and industries and promote energy-efficient equipment. These concerns are being addressed by numerous researchers in a number of ways. The more dependable, economical, ecologically beneficial, and widespread alternative strategy is to support renewable energy systems and related technologies. The construction of various hybrids of renewable energy sources has received a lot of attention in an effort to enhance long-term energy supply systems. Hybrid renewable energy sources (HRES) are systems that are reliable, CO2-emission-free, and an effective solution for minimizing dependency on one renewable resource, which is important in areas where natural resources are limited [4]. In view of [5,6], the integration of renewable energy resources is an emission-free solution for energy generation that allows energy supply in a district's topography and also acts as a steady potential energy resource for isolated generation applications. The renewable energy capacity indicated includes large-scale wind, solar, and home photovoltaic (PV) systems. Most residential PV systems are grid-connected, meaning the output receives excess electricity from the grid during the day and sends out power at night. HRES can be utilized independently for each home or in microgrids (MGs), which link a number of residences to create a small power grid in outlying areas where grid expansion is impractical [7,8]. The second method is gaining traction in rural areas and islands [9] because it is cost-effective and can be used alternatively in areas where power infrastructure upgrades are prohibitively costly and fuel transportation is problematic [10]. Many researchers have carried out research to develop hybrids from diverse renewable energy sources in order to improve long-term energy supply systems. According to studies on geographic information systems (GIS), the global population of islands is projected to be over 740 million [11]. Energy consumption has risen in recent years in islands and isolated regions, making reliance on fossil fuels uneconomical. As a result, standalone HRES and RES are viable long-term solutions for clean and cost-effective electricity for expanding populations and enterprises in remote areas and islands [12–14].

Since RES generates the majority of its energy from the environment, it reflects the environment's intermittent character. A significant disadvantage of wind and solar energy is how dependent they are on the environment. However, this issue may be solved by creating an HRES, which combines two or more energy sources with a backup unit [15]. HRES can be used with elements like wind and sunlight that complement each other. Moreover, energy storage systems may be combined with conventional energy sources like diesel generators (ESS). HRES can give a certain application a more cost-effective and steady electrical supply [16,17]. The high initial cost, rising cost of maintenance, fluctuating rates, and depreciation are key difficulties involved with hybrid systems [18]. In addition, HRES design is influenced by the availability of energy sources and site characteristics, as well as by technological and societal limits [19–21], which affect the system's power generation arrangements and total energy production cost.

The appropriate size combination is crucial in this scenario for providing enhanced reliability at the lowest cost. Determining the optimal design of HRES is a difficult undertaking because it is based on data from energy sources, technical specifications, ambient conditions, and load patterns [22]. HRES models, configurations, sizing, and optimization procedures have been investigated for a variety of locations and constraints [22–26]. Because solar and wind hybrid systems work well together [27], the majority of the studies have utilized them. Methodologies for optimizing the size of solar and wind hybrid systems have been combined, resulting in greater precision in optimization and control approaches in both grid-connected and stand-alone HRES [28–32]. Novel single-algorithms, hybrid algorithms, and software tools designed for grid-connected or remote sites and islands, as well as critical performance comparisons for all solar and wind hybrid system scaling, are among these techniques. Research focused on the application of artificial intelligence techniques in scaling HRES has highlighted a few discrete artificial algorithms for standalone and grid-connected applications [33,34]. Integration settings, storage system options, size approaches, and independent HRES control and management were key areas of interest [35]. In [36], the author presented the case study of sustainable energy production from municipal solid waste in Oman. In [37], authors presented the study related to the challenges towards renewable energy production the Arabian Gulf region. Reference [38] provides a discussion of the optimal design for several artificial single-algorithms and software tools, as well as a number of hybrid combinations. For independent and gridconnected applications, Upadhyay and Sharma [39] explored the size of various hybrid system combinations using both artificial and conventional sizing methodologies. The author of [40] mainly discussed multi-objective optimization approaches for hybrid energy systems using energy sources from fuel cells, the sun, and wind. The author of [41] concentrated on employing artificial optimization approaches to examine, control, and model HREs. Based on a series of probable accessibility, cost effectiveness, and emission-free environmental evaluation findings, the solar-biomass hybrid system is acknowledged and generally acceptable among various forms of HRES [42-44]. The use of solar and local biomass in hybrid systems maximizes the use of both. It may be possible to enhance the local energy structure by implementing these renewable energy technologies. In hybrid solar-biomass energy systems, the majority of the biomass subsystems directly absorb either forest biomass or agricultural waste.

A renewable energy generation system based on biogas and solar PV was described by Tazvinga and Dzobo [45]. The major goal was to boost a solar PV system's efficiency throughout the day and also include battery storage and a biogas generator to make up for its unpredictability. According to [46], the decentralized electricity supply based on renewable energy has traditionally been viewed as a single technology with a finite quantity of supply to satisfy essential needs. The current study's objective is to combine solar energy and biogas to provide electricity for an off-grid, rural community as an example. According to Rahmana et al. [47], solar and biogas are insufficient to cover both thermal (cooking) and electric load requirements. Biogas and hybrid solar energy applications are rather limited, despite their appealing potential. It is critical to examine the economic merits of these two resources, as well as their aptitudes to cope with needs and physical limitations, in order to enable their integration into rural energy planning and stimulate distribution by fully maximizing their potential. The maximum potential of solar PV energy and biogas may be used for solar PV-biogas hybrid power generation. Ansori and Yunitasari [48] recently explained how to electrify rural areas using a solar PV-biogas hybrid power generation system. The literature does not critically compare the various sizing optimization strategies' efficacy despite covering a wide spectrum of sizing optimization and a thorough study that included the most recent single and hybrid sizing optimization methodologies and software tools. A comparison of the efficiency of freestanding hybrid solar and wind systems for remote locations and islands has not yet been published. HRES are of potential use, especially for standalone systems, which are designed for remote and island areas as well as grid-connected systems for unreliable national grid demands. Because of this, the current study also intends to give a thorough review of recent advancements in single algorithms, hybrid algorithms, and software tools for the ideal size of HRES and evaluation characteristics including economic, reliability, environmental, and social considerations. Furthermore, this research assesses the size optimization methodologies employed by various researchers, and it has been thoroughly reviewed for standalone and grid-connected HRES with various energy sources and storage systems.

This paper has been structured into the following sections: Section 1 contains a detailed description of the components of a hybrid renewable energy system; Section 2 contains the paper's contribution; Section 3 contains the design parameters for a hybrid renewable energy system; Section 4 contains energy production unit sizing optimization; Section 5 contains energy storage system integrations on HRES plants; and Sections 6 and 7 contain the discussion, recommendations, and conclusions.

# Contribution of the Paper

This paper concentrated on hybrid renewable energy systems and their optimal sizes, which can operate as either national grid-connected or remote electrification systems. Considering the limitations and challenges identified in the above analysis, the contribution is made in the following ways:

- A description of the problem of selecting the best size for hybrid renewable energy systems
- Examining the present state of the art in hybrid renewable energy systems and the
  optimal size in relation to economic issues
- Organizing and categorizing existing research on the appropriate sizing of hybrid renewable energy systems with energy storage systems
- Identifying current technical challenges with reference to the optimal sizing of hybrid renewable energy systems with energy storage systems
- On the basis of numerous study fields, a full-scale constructive analysis of potential optimal sizing strategies and optimization methodologies was examined, highlighting objectives, major discoveries, and research gaps.
- Future research trends in the appropriate sizing of hybrid renewable energy systems with energy storage systems are likely to emerge.

#### 2. Overview of the Review Procedure

The review procedure for this research is depicted in Figure 1. There were four primary steps in completing this study. In order to identify the problem, the first step was an examination of feasibility constraints, components, objective functions, and methods. The second step analyzed and classified existing research on the topic using significant variables such as component, goal function, and approach. Shortcomings in this research were emphasized and thoroughly spelled out. Furthermore, the third step identified and discussed recent developments accordingly. The process's fourth and final step discussed future trends in optimal component planning for remote location power supply.



Figure 1. General approach for reviewing the study on HRES system sizing.

#### 3. Overview of HRES System Optimal Sizing

Some of the optimal sizing challenges associated with HRES systems are estimating system components with the most capacity and at the same time considering feasibility and reliability constraints. It's worth noting that the HRES grids are expected to be implemented

in this analysis, and only optimal generating and storage unit sizing is considered by using optimization methods [49,50]. Such is the case where HRES networks are typically built and developed by governments. Consequently, the distribution of grid installation on HRES systems lacks sufficient data for cost analysis. In addition, generation and storage units are generally located near rural locations, and the HRES grid has a far lower cost than traditional power networks [14,46,51]. Figure 2 depicts a generic technique for HRES system sizing optimization. The system's input data was used to start the optimal sizing algorithms for HRES system design. The HRES system setup was then defined. The sizing problem was stated using the optimization algorithm. In the next step, the HRES system's functionality was assessed. After the HRES system became operational, the feasibility limitations were checked for satisfactory results. The objective function was then calculated to complete the optimization problem, provided all the constraints had been met.



Figure 2. A general approach for optimal HRES system sizing.

#### 3.1. Hybrid Renewable Energy System Components

HRES has higher upfront costs, regional limitations, and a high degree of intermittency [52]. ESS is required to tackle the intermittency issue, even if the cost of HRES is decreasing [53]. However, the ESS cost is highly significant, especially when large-scale renewable power plants require a lot of capacity. For a cost-effective and ecologically friendly system, a hybrid diesel generator/HRES/ESS combination is recommended. A multi-component hybrid remote area electrification system, on the other hand, is a complex system that necessitates careful planning. In order to produce a reliable, cost-effective system, the concept of optimal planning is paramount. A hybrid renewable system [49,52,53] is the most cost-effective way to store and use natural power without interruption. Due to their dependability and cost-effectiveness in supplying energy to rural and remote areas, researchers have increasingly focused their attention on HRES integrated with ESS. Several studies [54–58] have examined resource utilization and techno-economic performance. Two or more renewable energy generation units, a backup fuel cell power generation unit that is optional, power conditioning units, and a storage unit are all components of an HRES production configuration system [59–63]. The most common schematic diagram of an HRES plant is shown in Figure 3, in which the load is fed first and foremost by solar and wind generators, with the biogas generator serving as a backup. The battery ensures power flow balance in the system as well as optimization.



Figure 3. Schematic diagram of an HRES system.

Components using fossil fuels to provide energy, such as diesel or gas generators, contribute considerably to greenhouse gas emissions. A range of renewable energy components has recently become available that can be incorporated with distant area electrification and national grid interconnection systems. The most readily available and appropriate components for far-flung electrical and national grid interconnected systems include solar PV, wind turbines, hydropower, and biogas generators. Their use, however, is strongly dependent on the geographic location of the research site [64–66]. Due to the abundance of biomass in rural regions, biogas producers will attract greater attention in the near future [66]. Figure 4 presented the system components in HRES systems.



Figure 4. System components in HRES systems.

#### 3.2. Design Parameters of HRES System

When constructing a hybrid renewable energy system, the most important elements to consider are cost and reliability. These variables are related to emissions and technological challenges. The type of objective function utilized was based on the type of investiga-

tion. Often times, economic objectives take precedence. If the project's budget is limited, reliability becomes a major problem. Emissions have drawn a lot of attention in several situations. Because the objectives are so different, optimal sizing in hybrid renewable energy systems can be achieved by using optimization techniques to solve a single-objective or multi-objective optimization issue. A compromise between the objective functions is required for multi-objective issue solutions expressed as Pareto fronts [67].

As shown in Figure 5, the different categories of objective functions are presented. Nowadays, most researchers give priority to economic factors, then reliability factors, and end with technical and emission considerations. Each of the above objective functions' categories are explained in detail below.



Figure 5. HRES system sizing based on objective functions.

### 3.2.1. Objective Functions of Finance

Financial goals include the net present cost (NPC), levelized cost of energy (LCOE), total annual cost (TAC), simple payback period (SPP), and internal rate of return (IRR). The NPC of a diesel generator is calculated by adding up all current capital, maintenance, replacement, salvage, and fuel consumption costs [68]. The capital recovery factor is multiplied by the NPC over the system's yearly energy consumption to determine the LCOE [69]. To calculate TAC, yearly construction and maintenance costs are compounded by annual fuel prices [70]. The SPP measures how long it will take for yearly profits to cover component capital expenses [71]. The discount rate at which the net present value (NPV) of all future cash flows is zero is known as the IRR [72]. The mathematical formulation of each economic objective function for the hybrid renewable energy system size is as follows:

(a) NPC: The present value of all benefits and costs that will occur throughout the project's lifetime is known as the net present cost [73].

$$F_1 = Minf(NPC) = NPC_k + NPC_f$$
(1)

$$NPC_k = PC_C + PC_m + PC_r - PC_s$$
<sup>(2)</sup>

$$NPC_f = \left(\frac{(1+r)^n - 1}{r(1+r)^n}\right) \times \left(\sum_{t=1}^T \left(f(t).C_f\right)\right)$$
(3)

(b) LCOE: It represents the system's entire yearly cost per kWh of useable electrical energy [74].

$$F_{2} = Minf(LCOE) = \frac{NPC_{k} + NPC_{f}}{E_{p}} \times \frac{r(1+r)^{n}}{(1+r)^{n} - 1}$$
(4)

(c) TAC: It is the annualized cost of all power system components, which includes replacement and fuel expenses in addition to capital, operating, and maintenance costs [75].

$$F_3 = Minf(TAC) = \sum_{t=1}^{T} \left( f(t).C_f \right) + AC_k$$
(5)

(d) SPP: It is the amount of time needed to recoup an investment's cost [73].

$$F_4 = Minf(SPP) = \frac{PC_c}{AP}$$
(6)

(e) IRR: In a discounted cash flow analysis, it is a discount rate that sets the net present value (NPV) of all cash flows to zero [76].

$$F_5 = Maxf(IRR) \tag{7}$$

$$-PC_{c} + \sum_{y=1}^{Y} M_{Y} \times (IRR)^{y} = 0$$
(8)

#### 3.2.2. Objective Functions of Reliability Evaluation

The following are some of the most common measurements and target functions for HRES optimal sizing dependability.

- 1. Loss of power supply probability (LPSP)
- 2. Expected energy not supplied (EENS)
- 3. Loss of load expectation (LOLE),
- 4. Loss of energy expectation (LOEE)

Further, the system average interruption frequency index (SAIFI) and system average interruption length index (SAIDI) are two other dependability indices that have received less attention for optimal sizing of HRES. The likelihood of an unmet load over the whole energy demand of a grid-connected or stand-alone hybrid renewable energy system is known as the LPSP [77]. The EENS is the energy that is supposed to be provided by a hybrid renewable energy system but is not [78]. The LOLE, also known as the loss of load probability (LOLP), is the number of hours per year that the energy exceeds the capacity of the HRE generation system [79]. The LOEE [80] stands for the total energy not delivered by the grid-connected or stand-alone hybrid renewable energy system. Over the course of a year in the HRES project, SAIFI can be defined as the average number of times a client witnesses power outages. Throughout the life cycle of the project, the SAIDI index measures the total average customer's interruption time. For hybrid renewable energy systems, we explain the mathematical calculation of reliability objective functions.

$$F_{1} = Minf(LPSP) = \frac{E_{p} + E_{d} + E_{b,ch} - E_{re} - E_{f} - E_{b,dis}}{E_{p}}$$
(9)

$$F_2 = Minf(EENS) = \sum_{t=1}^{T} L_P D_P$$
(10)

$$F_3 = Minf(LOLE) = \sum_{t=1Ses}^{T} \sum F_s T_s$$
(11)

$$F_4 = Minf(LOEE) = E_p + E_d + E_{b,ch} - E_{re} - E_f - E_{b,dis}$$
(12)

$$F_5 = Minf(SAIFI) = \frac{\sum \lambda_i N_i}{\sum N_i}$$
(13)

$$F_6 = Minf(SAIDI) = \frac{\sum U_i N_i}{\sum N_i}$$
(14)

3.2.3. Objective Functions of Emission and Technical

The following are the other groups of objective functions:

- 1. Renewable factor (RF)
- 2. Carbon emission (CE)
- 3. Battery longevity (BL)
- 4. Customer comfort level (CCL)
- 5. Discharged energy (DE)

The RF shows how much of the energy demand is fulfilled by HRES [81]. The CE represents the total quantity of  $CO_2$  emitted by the envisaged HRES system over the project's duration [82]. The BL is the battery's lifespan in HRES that has been shortened due to deterioration. In order to avoid battery damage and thus increase battery lifetime, a proper installation plan should be developed. The mathematical formulas for emission and technical objective functions are offered in Equations (15)–(19) in the HRES optimal sizing issue. However, the demand response solution for this study has an impact on CCL formulation. The number of hours required to achieve the greatest CCL might be decreased, for instance, if load shifting is considered. The inverter management system, which reduces power fluctuations and provides a consistent power supply, is taken into consideration when calculating the EFR. For the optimum size of a hybrid renewable energy system, the emission and technical objective functions are mathematically formulated.

$$F_1 = M \inf(RF) = \left(1 - \frac{E_f}{E_p}\right) \times 100 \tag{15}$$

$$F_2 = Minf(CE) = \alpha + \beta \sum_{t=1}^{T} P_f(t) + \gamma \left(\sum_{t=1}^{T} P_f(t)\right)^2$$
(16)

$$F_3 = Maxf(BL) = 1 - D_b \tag{17}$$

$$F_4 = Maxf(CCL) \tag{18}$$

$$F_{5} = Minf(DE) = E_{re} + E_{f} + E_{b,dis} - E_{p} - E_{b,ch}$$
(19)

#### 3.3. Consideration of Feasibility Constraints

There are two types of feasibility restrictions for hybrid renewable energy system sizing. These include (1) component-related restrictions and (2) system-level technological constraints. The feasibility constraints on remote area electrification and the optimal sizing difficulties of the national grid interconnection system are shown in Figure 6.



Figure 6. Constraints on optimal sizing of HRES.

#### 4. Sizing Optimization of Energy Production Unit

References [83–85] provide a comprehensive examination of the critical unit sizing difficulty for hybrid renewable energy systems. It is a technique for estimating hybrid system component sizes that also lowers system costs and increases system reliability [86]. Oversizing can raise the cost of the system, whereas under-sizing could lead to a power supply breakdown or inadequate power being delivered to the load. There are several techniques for sizing the appropriate hybrid renewable energy system. Among the various options, there are two that are more common and commonly used.

- (1) Simulation and optimization software
- (2) Meta-heuristic optimization techniques

#### 4.1. Simulation and Optimization Software

Simulation tools are the most widely utilized instruments for assessing the performance of hybrid systems. By evaluating the efficiency and cost of energy generation of various system configurations using computer simulations, the ideal design may be found. Just a few of the software tools that may be used to create hybrid systems are HOMER, HYBRID2, HOGA, and HYBRIDS. The HOMER (Hybrid Optimization Model for Electric Renewables) tool from the National Renewable Energy Laboratory is easy to use. It assesses hybrid renewable energy using hourly simulations and environmental data, then optimizes the system using the net present value. Many studies utilizing HOMER [87] have been undertaken on the best design of hybrid renewable energy systems without ESS. HOMER was used to optimize a diesel generator-PV-Wind-battery hybrid [88], a PV-Wind hybrid [89], a mini-hydro-wind hybrid [90], a solar-biomass hybrid [91], and a hydro-windsolar hybrid [92]. The PV-WT-DG-biogas system was sized by HOMER for a community service power application since it is simple to operate [93]. In [94], the researcher optimizes the design of a biogas generator for a hybrid remote area electrical system in a distant village with a WT-PV-DG. Due to the intermittent nature of solar irradiation and wind speed, this hybrid system was unable to deliver a steady supply for the connected demands. To address the dependability issue, hybrid renewable energy systems should incorporate energy storage technologies.

Numerous academics are looking at how to best construct hybrid renewable power plants using HOMER software, integrating energy storage devices with remote region electrification systems, and linking the national grid system. The ideal HRES system size determined by the HOMER software optimization tool is summarized in Table 1. The optimal WT-PV-DG-BES size with the optimal levied energy cost was developed for use in islands, rural remote locations, and off-grid communities [95–98]. The PV-WT-BES system is the most commonly discussed method for remote area electrification and grid interconnection [99]. Other technologies used in clean distant regions for electricity and national grid connectivity, in addition to PV and WT, include fuel cells, super- and ultracapacitors, and pumped hydro. Long-term and short-term energy storage systems are divided into two categories. Long-term energy storage includes pumped hydro and fuel cell systems, whereas short-term storage includes batteries and ultra- and super-capacitors. The majority of studies recommended combining solar with long-term energy storage systems on a large scale [100]. HESS has been extensively researched for remote electrification and national grid connectivity systems. In South Africa, a hybrid FC-SC HESS with a photovoltaic system was used for commercial remote loading [101]. With a hybrid PV and WT system, the optimal size was found to be a combination of BES and FC [102]. To create a clean hybrid system with more electricity supply flexibility, a biogas generating unit has been combined with a PV-WT-BES system [103]. In [104], an agricultural farm's biomass-biogas system was sized to perfection. HOMER [105] was also used to examine the use of biogas-producing units in conjunction with hydropower in clean remote area electricity systems as well as national grid connectivity systems.

Ref.	Decision Variable	<b>Optimization</b> Method	<b>Objective Function</b>	Design Constraints	Electricity Tariff
[106]	Wind/PV/FC/BES	Homer Pro	NPC, COE, initial investment cost, and operating cost	Power balance and budget	Time of use
[107]	PV/wind/Biogas/FC	HOMER Pro	COE and NPC	Power balance and budget	Time of use
[108]	PV/Wind /BES/DG	HOMER	NPC, COE, and RF	Load demand, diesel fuel price, project lifetime, and interest rate	Time of use
[109]	PV/Wind/ BES/DG	HOMER	NPC and LCOE	Electricity production, emission, operating cost, fuel consumption	Time of use
[110]	DG/PV /Micro hydro	HOMER	Operating costs and return on capital	Power balance and budget	Stepwise Real time pricing
[111]	PV/Biomass /BES/DG	HOMER	LCOE	Required electrical load and available energy resources	Time of use
[112]	Wind/DG/PV/BES	HOMER	NPC	Single criterion-total net present cost	Not specified
[113]	PV/Wind/ DG/BES	HOMER PRO	NPC and COE	Capital cost, energy generated, excess energy, unmet load, life cycle emission, renewable penetration	Time of use
[114]	PV/wind /BES	HOMER, QRod <sup>TM</sup> & PROSPER <sup>TM</sup>	NPC and LCOE	Load demand, capital cost, available energy resources, and energy generated	Not specified
[115]	PV/DG/BES	HOMER Pro	COE and NPC	Load demand, capital cost, available energy resources, and configuration of RES	Time-of-use

Table 1. Optimal sizing of HRES systems by using HOMER software optimization tool.

The essential strength of each study project is the authors' strategy to deal with the community's issues with a shortage of power, as mentioned in the review articles in Table 1. According to the decision variables, every hybrid system, excluding references to the community's issues with a shortage of power, as mentioned in the review articles in Table 1, according to the decision variables, has a diesel generator. Diesel generators are not only not environmentally friendly, but they are also not economically viable. Besides, BES is not completely economically feasible. A future study should take all of the aforementioned features and drawbacks into account and make sure to address them in the brand-new, exclusively green hybrid system configuration that will be used on the system. Additionally, to deal with both the objective function and the constraints, the future researcher should handle the issues utilizing metaheuristic optimization approaches.

#### 4.2. Meta-Heuristic Optimization Techniques

It is critical for designers to develop a practical optimization technique for determining the best system size and configuration for hybrid renewable power plants. For constructing a hybrid renewable energy system, there are numerous optimization techniques available, the most popular and accurate of which is metaheuristic optimization. Metaheuristic approaches are commonly used to achieve appropriate HRES sizing. Existing metaheuristic technique studies are categorized into one or more objective optimization studies. The reference number, decision variables, optimization methods, objective function, design restrictions, and electricity tariff of extant metaheuristic studies on single-objective and multi-objective optimal design of HRES are shown in Table 2. Metaheuristic approaches were used to size the hybrid PV-WT-DG-BES system, which lowers the cost of energy production [116]. The LPSP was employed as a constraint to increase dependability [117–120]. The number of components and the power balance between generation and consumption were the most commonly used feasibility restrictions. However, the researchers suggested a system with a PV-WT-DG-BES that was both cost- and size-optimized [121,122]. Numerous methods were examined for improving the RF, unit commitment, and proportion of renewable energy [123–127]. The aforementioned studies were peer-reviewed and have only one objective.

Furthermore, Table 2 demonstrates that the reference numbers represent the singleobjective optimal design of a hybrid renewable power system, whereas the remaining references represent the multi-objective optimal design of a hybrid renewable power system. Many researchers have economic goals as their first priority. Furthermore, objective functions linked to pollution and reliability were the most commonly used. The researcher also recommended that three objective functions, such as renewable factor (RF), carbon emission (CE), and life cycle cost (LCC), be evaluated jointly to build a hybrid renewable power plant ideally [127]. However, because CE and RF belong to the same type of emissionminimization target functions, it is unnecessary to consider them for optimal sizing. The researcher in [128] took into account three objective functions as well as new limitations such as the WT hub height and PV tilt angle.

As stated in the previous two paragraphs, metaheuristic approaches are used as a single objective and a multi-objective for the optimal design of a hybrid renewable power plant in a clean energy production system. However, in the best design of a hybrid renewable power plant, the emission objective functions are removed as a result of the limited diesel generators available for clean energy production schemes. In [129], the researcher designed a WT-PV-BES system optimized in a group of twenty households, resulting in cost-effective and emission-free energy generation with reduced energy costs. In other studies, such as [130], four distinct algorithms were utilized to examine the performance of the metaheuristic algorithm for optimal sizing of hybrid renewable power plants. Using a PV-thermal system, [85] evaluated the supply of thermal loads in addition to the electric loads. Furthermore, a natural gas backup boiler was optimized in addition to the renewable system [131]. Both the hybrid grey wolf optimizer-sine cosine approach and the modified bee algorithm were used to determine the ideal HRES size [132,133]. Particle swarm optimization was used to improve the PV-WT-BES and biogas-PV-WT systems [134,135]. A PV-WT-PHS system was created in [136] to provide loads in a seaside village. Due to the abundance of water for PHS, such a system is extremely effective in coastal regions. The method with the most popular and greatest applications was the particle swarm optimization technique, which had several objectives. The current research considers objective functions such as volatility [137] and minimization of the total energy cost and loss of power supply probability [138]. The most commonly studied HRES configurations were WT-PV-FC [139], WT-PV-PHS [140], WT-PV-BES [141], WT-PV-FC [142], and PV-FC-BES [143]. However, reference [144] presented an improved two-component PV-BES system.

Ref.	Decision Variable	Optimization Method	<b>Objective Function</b>	Design Constraints	Electricity Tariff
[145]	PV/Wind/ FC	Hybrid firefly-harmony search optimization	NPC	Power balance and techno-economics	Time of use
[146]	PV/Wind/ FC	Hybrid grey wolf optimizer-sine cosine algorithm	LCC		Time of use
[147]	PV/wind/ BES/PHS	Four algorithms	NPC	Number of components, battery's energy and SOC	Time of use
[148]	PV- Thermal/WT/micro- turbine/ EES/Thermal energy storage/Natural gas boiler	Evolutionary PSO	TAC	Investment, replacement, fuel, and operation and maintenance costs. Energy management system prioritizes the application	Time of use
[149]	PV/wind/ /PHS	Genetic algorithm	LPSP	Power balance and techno-economics	Time of use
[150]	PV/Wind/ PHSS	Whale optimization algorithm (WOA)	COE	Power balance and budget	Time of use
[151]	PV/wind/ BES	Crow and particle swarm as a hybrid	Reduction of energy production cost	Distribution of energy supply-demand planning	Time of use
[152]	PV/Wind/ Biogas/DG/BES	Hybrid PSO-GWO	COE and LPSP	Optimal configuration according to the cost	Time of use
[153]	Wind/PV /BES	Genetic algorithm-III (NSGA-III)	Total cost, end-user satisfaction loss, and tie-line power fluctuation	Power balance and budget	Time-of-use
[154]	PV/Wind/FC/BES	Proximal policy optimization (PPO)	Overall economic cost saving and carbon emission reduction	Power balance and techno-economics	Time of use
[155]	PV/Wind/ BES/DG	Multi-objective multi-verse optimization (MOMVO)	COE, RF, and LPSP	Required electrical load and the techno-economic feasibility	Time-of-use
[156]	Wind/PV/FC/ BES	WOA	COE, NPC, and LPSP	Produce an adequate electrical supply to the load demand with low cost	Time-of-use
[157]	PV/Bio-waste /FC	WOA	NPC) and LPSP	Electrical load, optimal configuration, and techno-economic feasibility	Not specified
[158]	WT/PV/Biomass/Pump- Hydro	WOA	COE and LPSP	Reliability and operational constraints	Time of use
[159]	PV/Wind/ BES/DG	NSGA-II	NPC, COE, and CO2 emissions	Power balance and techno-economics	Time of use

**Table 2.** Single and multi-objective capacity optimization for HRES with meta-heuristic optimization techniques.

The researchers' approach to addressing the community's problems with a lack of electricity, as noted in the review articles in Table 2, is the key strength of each research project. Regarding its economic and environmental implications, references [147], [149], [150], [158], and [159] are significantly superior to the other articles provided. Diesel generators are not both economically and environmentally feasible in a hybrid system. BES is also totally uneconomically viable. Future research should include the aforementioned benefits and drawbacks and make sure to address them specifically in the green hybrid system configuration that will be employed on the system.

# 5. Application of the Integration of Energy Storage System in HRE Plants

The main disadvantage of using renewable energy is limited to the fact that it cannot deliver reliable electricity as a result of its intermittent nature [160]. Energy storage systems (ESSs) are the most effective way to store power during off-peak hours and supply energy during peak hours [161]. For the load to get an uninterrupted supply of power, storage technology is crucial and required [162]. Alternatives for energy storage in HRES include CAES, PHS, FWES, SC, SMES, and BES.

These devices are often employed in large-scale networks, which require significant capital. However, they can be used to ensure a consistent energy supply during worse HRES conditions [163]. One of the most widely utilized ESSs is the battery energy storage system (BESS) [164]. Consequently, combining HRES and BESS is a potential on- and off-grid solution, not just in India but internationally. MGs have become more popular in recent years as people have become more interested in using them in power distribution networks using small-scale HRES. Moreover, the microgrid idea has been regarded as a superior alternative for countryside electrification, and many hybrid MG designs for HRES have been given in the literature [165,166]. MGs are frequently recognized as the most dependable, consistent, economical, and environmentally friendly energy sources. An MG is a standalone electrical system that may provide electricity to a household or community. The abundance of HRES makes utilizing these sources as a remote area electrification option a strong prospect [167]. Hybrid configurations, particularly MGs or HRES, can combine energy conversion systems such as PV and wind turbines. These hybrid topologies will reduce generation, investment, and storage system size fluctuations and simultaneously boost system reliability and performance [168]. As a result, the ESSs offer backup energy when the HRES' output power fluctuates. The HRES's resilience is improved, and total expenses are reduced by the integration of the ESS [169]. Table 3 surmises the optimal sizing of HRES with ESS systems by using meta-heuristic optimization techniques. Furthermore, the HRES' reliability is ensured by its continuous power delivery to the load [166]. Using a diesel generator (DG) ensures uninterrupted power loading during HRES. Renewable energy-based MGs can run in island mode, reducing reliance on fossil fuels. It also provides significant economic and environmental benefits [169]. For this reason, localized HRES integrated with ESS is a better solution for satisfying the energy demands of load centers in a reliable way.

Table 3. Optimal sizing of HRES with ESS by using meta-heuristic optimization techniques.

Ref.	Decision Variable	Technique	<b>Objective Function</b>	Constraints	Electricity Tariff
[170]	Wind/PV/ BES	Firefly-inspired algorithm	COE	Energy of battery, number of components, and load dissatisfaction rate	Time of use
[171]	Biogas/PHES /PV/BES	Water cycle algorithm	NPC	LPSP, number of components, SOC, upper reservoir volume	Time of use
[172]	WT-PV-FC	Artificial bee swarm optimization	LCC and LPSP evaluation	Load interruption probability, number of components, energy at tank	Time of use
[173]	PV-BES	Mutation adaptive differential evolution	LCC, LOLP & LCOE	SOC	Time of use and stepwise real time pricing
[174]	PV/wind/BES	Multi-objective grey wolf algorithm	COE, LPSP, DE	SOC	Time of use
[175]	PV-WT-BES-PHS	Multi-objective grey wolf A.	COE, LPSP	Energy of battery and pump-hydro storage	Time of use

The researchers' approach to addressing the community's power problems with intermittent forms of electricity production like solar and wind, as noted in the review articles in Table 2, is the key strength of each research project. In terms of economic implications, BES outperforms the other ES system significantly. In addition, PHS, FC, and BES do not respond quickly when there is a peak load occurring in milliseconds, so researchers should include fast response ES systems like SMES and FWES on the intermittent HRES. Future research should include the aforementioned benefits and drawbacks and make sure to address them specifically in the green hybrid system configuration that will be employed on the system.

The technical and financial aspects of various energy storage systems used for renewable and hybrid energy alternatives are shown in Table 4. People often think that dispatchability, efficiency, durability, availability, quick response time, energy capital cost, and so on are the most important things for a storage system to have. In contrast to battery storage technology, which can only make 0 to 40 MW of energy available, the PHES can make 100 to 5000 MW of energy available. Compared to thermal and chemical energy storage methods, it is more efficient. The PHES's longer lifespan than any other storage system is one of its best qualities. When compared to other storage methods, PHES has a low capital cost for energy. According to Hino and Lejeune [176], PHES plants have quick start-up and shut-down times, quick load changes, the ability to handle frequency changes, and stable voltage. Nazari et al. [177] discuss that PHES systems are useful tools for making sure that there is always power. In general, PHES has a much lower LCOE than other ways to store energy. Based on these qualities, it's clear that the PHES system is better than all other storage systems. In [178], Zhang et al. proposed the Mo6+–P5+ co-doped Li2ZnTi3O8 anode for Li-storage in a wide temperature range and applications in LiNi0.5Mn1.5O4/Li2ZnTi3O8 full cells. In [179], Chen presented research on the use of digital twin technology for collaborative innovation of important common technologies in the new energy vehicle industry. Future low-carbon and zero-carbon fuels for marine engines were studied in [180] from the perspective of thermal efficiency. In [181], Liu et al. conducted a numerical analysis of the ammonia combustion and emission properties in a low-speed two-stroke marine engine. A thorough analysis of smart distribution network situation awareness for high-quality operation and maintenance was published by Ge et al. in [182]. Li et al. presented the digital economy's driving mechanism in [183] based on a regulation algorithm for the growth of low-carbon sectors. The improved algorithm of drift compensation for olfactory sensors was presented by Lu et al. in [184]. The semi-supervised extreme learning machine approach based on the new weighted kernel for machine scent was introduced by Dang et al. in [185]. The asymmetric encoder-decoder model for Zn-ion battery lifetime prediction was introduced by Lu et al. in [186].

Table 4. Economical and technical criteria for various energy storage technologies.

ES Technology	Capital Cost (\$/kW)	Power Rating (MW)	Energy Density (Wh/kg)	Power Density (W/kg)	Life Cycle	Response Time	Life Span	Efficiency (%)	Ref.
Lead Acid Battery	300-600	0–40	24-45	180	1500-2000	5–10 ms	3–12	70–90	[187] [188]
SC	100-300	0.01–1	0.1–5	800-2000	100,000+	<5 ms	10–20	85–95	[189] [190]
FWES	110-330	0.01–10	10–30	400-1500	10,000– 100,000	seconds	15–20	70–95	[191] [192]
FC	500-10,000	0.001–50	300-1200	500+	20,000+	min	5–20	20–50	[193] [194]
CAES	400-800	5–300	30–60	-	8000-12,000	min	20–40	70	[195] [196]
SMES	200-300	0.1–10	0.5–5	500-2000	100,000+	<5 ms	20–30	90–98	[197] [198]
PHS	600-4300	100-5000	0.5–1.5	-	10,000– 30,000	min	30–60	65–85	[199] [200] [201]

#### 6. Discussion and Recommendations

With increased electricity demands and the intermittent nature of single renewable energy sources, it is increasingly difficult to provide dependable power to linked loads. By minimizing maintenance expenses, which decrease the system's overall operating costs, an effective and long-lasting energy storage technology can address the issue of HRES's intermittent nature. Simultaneously, hybridization, in conjunction with energy storage technologies, can address the intermittent nature of HRES. Energy storage possibilities in HRES include the following options:

- 1. Compressed air energy storage (CAES)
- 2. Pumped hydro energy storage (PHS)
- 3. Hydrogen fuel cells (FC)
- 4. Flywheels
- 5. Super capacitors (SC)
- 6. Superconducting magnetic energy storage (SMES)
- 7. Battery banks (BB)

PHS units have different advantages over CAES, SC, SMES, FC, flywheels, and BB units, such as lower prices and less environmental impact.

PHS is the most effective way to store media among several options. It has several advantages over traditional energy storage systems, including fast response time, quick starting and stopping, ease of handling load changes, high-efficiency power supply at a base load power plant, and decreased discharge losses. In both experimental and computational studies, PHS has been used as a bulk energy medium for HRES by a number of researchers. Numerous studies have established the efficacy of PHS in various configurations. It is considered to be one of the most reliable and technologically possible off-grid as well as grid-connected HRES power sources for use in any electricity demand sector.

The solar-wind-PHS combo is considered a reflection of enormous solar and wind potential due to increasing installed capacity and peak demand and supply. The COE, LPSP, environmental impact, and payback period may all be decreased using the integrated system. Researchers have also found that combining a solar-biogas system with a PHS system provides advantages such as lower investment costs, improved operating performance, and smooth power generation. Furthermore, hybrid systems that are connected to the grid have the best COE in the majority of circumstances. The reasons can be summed up in the lower cost of kWh gained from the grid compared to the initial expenditures obtained from renewable energy sources. However, acceptable rate reductions in the initial prices of renewable energy have been observed in recent years.

PHS is now combined with PV-wind-biogas-based HRES for a continuous and stable power supply, with an internal combustion engine acting as a backup energy source. Hybrid-PHS configurations were also researched by several researchers.

The integrated system offers improved round-trip efficiency, enhanced power supply dependability, decreased revenue losses, cost savings, a low investment cost, maximum accessible energy, a greater life span, and fewer greenhouse gas emissions when compared to a battery and other storage systems. Prior studies have revealed that PHS and freshwater resources appear to be among the most practical HRES storage solutions.

The following suggestions have been provided to overcome the above challenges to optimal sizing of HRES adoption with ESS integration:

- The current state of HRES technology, interconnected with ESS, can address many of the issues that the prior technology had, such as reliability, efficiency, and capacity. The scope of this technology's ongoing development for future use in MG technology has been determined. Energy sizing, costing, safety, and effective management are increasingly the focus of research.
- For HRES and ESS system components to be sized optimally, intelligent techniques (meta-heuristic approaches) must be combined with the proper control settings, or more effective methods must be developed. It may be said that the hybrid GWO-PSO and WOA optimization strategies are the best for achieving the aim of an HRES combined with ESS that is dependable, economical, and environmentally benign.
- The components of renewable energy and the life cycle of storage devices are determined by the materials utilized. Capacity, energy and power density, life cycle, corrosiveness, and charging and discharging properties may all be significantly influenced by the materials. With better energy efficiency, reliability, and stability, a cost-effective long-term advanced technology can lead to the material selection of HRES and ESS in MG applications.

- To combine HRES with ESS and the current electrical power network, the power electronic interface (PEI) can be employed. Because it possesses the requisite organization for power conversion, PEI has a variety of features. Size, ripples, expense, flexibility, and efficiency are all shortcomings of the current PEI system.
- Sharing the power allows for the optimal distribution of power in the HRES with the ESS structure. PHS, FC, CAES, and Li-ion batteries are just a few of the ESS that can be modeled for large-scale integration. Thermal energy storage systems, SMESs, flywheels, and flow batteries all perform well for medium-scale energy management. A quality management system could be utilized to boost the overall efficiency and cost of present ESS management for HRES applications that have consistent and reliable quality.
- Different ESS technologies are quite large and expensive in terms of size and cost. An
  ESS that is too large is not appropriate. Installation and maintenance costs are included
  in the price. It also significantly contributes to storage permanence. Their integration
  can boost the storage system's capacity. Implementing a comprehensive energy storage
  policy would be a big issue for both renewable and conventional networks.
- In order to improve system stability and dependability and simultaneously lower
  power quality concerns, HRES requires an ESS that combines the traits of a high-power
  and a high-energy storage system. High-energy devices have a slower reaction with
  a longer duration, but high-power ESS devices benefit from rapid responses at high
  rates for a short period of time. The advantages of achieving excellent power quality
  with linked loads may be realized by combining these two kinds of ESSs.
- A predetermined operating policy should be implemented to assess the site's longterm viability. Many restrictions can be overcome with technological advancement. Transmission losses can be reduced by choosing the right PHS site. Additionally, PHS's integration with solar farms that are almost entirely self-sufficient will reduce transmission costs between the two businesses. In order to boost the new PHS's societal acceptance, it is important to spread awareness about the project's efficiency and viability as a source of power. Furthermore, community communication and consultation can help increase public interest. The success narratives of successfully completed projects must be shared with the public in order to raise awareness and recognition.
- The emission of greenhouse gases and other hazardous emissions decreases as the amount of energy supplied by renewable sources grows. Hybrid ESS can incorporate intermittent HRES into the power system, lowering fuel usage and hazardous emissions. Despite the fact that 100% renewable energy production is expensive, experts are working to lower installation and maintenance costs.

# 7. Conclusions

In this work, the state of the art for HRES system optimum sizing was investigated. The current research on the subject was divided into categories using HRES, optimization methods or software optimization, and single- or multi-objective problems. The most recent advancements in HRES integration with ESS system optimal sizing, as well as current issues, were reviewed. Future views were offered to scholars as a way of highlighting new research topics. The following are some of the main results of this review paper.

Based on many research findings, FITs for loads in grid-connected HRES should be implemented by sending surplus energy to the grid system. This increases the proportion of HRES that uses renewable energy resources. Consumers can therefore sell their excess energy to the grid through the FIT, which lowers electricity costs and generates revenue for the community.

To optimize the size of components based on economic, reliable, and emission functions, new meta-heuristic optimization approaches and software tools are needed.

Meta-heuristic optimization techniques are more efficient for sizing HRES. However, current software tools, such as the HOMER software family, are unable to address multiobjective issues. Additionally, demand-side management response systems are difficult to deploy with this software. As a result, software could be deployed, offering designers the flexibility to size HRES systems more efficiently.

Author Contributions: Conceptualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A., A.F.-L., I.A. and E.T.; software, T.F.A. and A.F.-L.; validation T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; formal analysis, T.F.A. and A.F.-L.; investigation, T.F.A. and A.F.-L.; resources, T.F.A., A.F.-L., I.A., E.T., A.A. and B.K.; methodology, T.F.A.; data curation, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; writing—original draft preparation, T.F.A., A.F.-L., I.A., E.T., A.A. and B.K.; methodology, T.F.A.; writing—review and editing, T.F.A., A.F.-L., I.A., E.T., A.A. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T., A.A. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; N.F.-L., I.A., E.T., A.A. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; visualization, T.F.A., A.F.-L., I.A., E.T. and B.K.; methodology, T.F.A.; project administration, T.F.A., A.F.-L., I.A., E.T., A.A., B.K. and C.L.R.V.; methodology, T.F.A., A.F.-L., I.A., A.A., and E.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Data will be available on request.

**Acknowledgments:** Authors thanks the support of the MIRET Scholarship program through project No: 614658-PANAF-1-2019-1-KE-PANAF-MOBAF.

Conflicts of Interest: The authors declare no conflict of interest.

#### Abbreviations

$AC_k$	Annual Cost of Components	MG	Micro-Grid
AP	Annualized Payment of HRES	$M_Y$	Net Cash Flow For The Year
CAES	Compressed Air Energy Storage	Ν	Project's Lifespan
C <sub>f</sub>	Fuel Price	$N_i$	Number Of Customers for Site
ĆOE	Cost Of Energy	NPC	Total Net Present Cost
Cr	Cost of Fuel	$NPC_{f}$	Fuel Usage
$D_b$	Battery Capacity Degradation From Charging/Discharging Cycles	$NPC_k$	Grid Integrated or Remote System Components
DG	Deiseal Generator	NPV	Net Present Value
$D_P$	Unmet Load Duration	$PC_C$	Capital Costs Present Value
$E_{b,ch}$	Total Energy Output from Charged Battery	$PC_m$	Maintenance Costs Present Value
$E_{b,dis}$	Total Energy Output of Discharged Battery	$PC_r$	Replacement Costs Present Value
$E_d$	Total Dumped Energy	$PC_s$	Salvation Costs Present Value
EENS	Expected Energy Not Supplied	$P_f$	Power Generated by the Diesel Generator
Ef	Diesel Generator Energy Output	PHS	Pumped Hydro Storage
É <sub>p</sub>	Overall Energy Demand	PV	Photo Voltaic
É <sub>re</sub>	Total Output From Renewable Energy Sources	R	Discount Rate
ESS	Energy Storage System	S	All Loss of Energy States
f	Fuel Utilization	SAIDI	System Average Interruption Length Index
$F_S$	Probability Of Meeting States	SAIFI	System Average Interruption Frequency Index
HOMER	Hybrid Optimization Model for Electric Renewable	SMES	Superconducting Magnetic Energy Storage
HRES	Hybrid Renewable Energy Sources	SPP	Simple Payback Period
IRR	Internal Rate Of Return	Т	Project Time Duration
LCC	Life Cycle Cost	TAC	Total Annualized Cost
LCOE	Levelized Cost of Energy	$T_S$	Loss of Load Duration
LOLE	Loss of Load Expectation	$U_i$	Duration Of Power Outage
LOEE	Loss of Energy Expectation	αβγ	Approximate Emission Coefficients
$L_P$	Average Yearly Load	$\lambda_i$	Rate of Power Interruption
LPSP	Loss of Power Supply Probability		-

#### References

- 1. Abaye, A.E. System Analysis and Optimization of photovoltaic–wind hybrid system. System 2018, 5, 197–201.
- Kuang, Y.; Zhang, Y.; Zhou, B.; Li, C.; Cao, Y.; Li, L.; Zeng, L. A review of renewable energy utilization in islands. *Renew. Sustain.* Energy Rev. 2016, 59, 504–513. [CrossRef]
- Ammari, C.; Belatrache, D.; Touhami, B.; Makhloufi, S. Sizing, optimization, control and energy management of hybrid renewable energy system—A review. *Energy Built Environ.* 2022, *3*, 399–411. [CrossRef]
- 4. Zebra, E.I.C.; van der Windt, H.J.; Nhumaio, G.; Faaij, A.P. A review of hybrid renewable energy systems in mini-grids for off-grid electrification in developing countries. *Renew. Sustain. Energy Rev.* **2021**, *144*, 111036. [CrossRef]
- 5. Semaoui, S.; Arab, A.H.; Bacha, S.; Azoui, B. The new strategy of energy management for a photovoltaic system without extra intended for remote-housing. *Sol. Energy* **2013**, *94*, 71–85. [CrossRef]

- Robitaille, M.; Agbossou, K.; Doumbia, M.L. Modeling of an islanding protection method for a hybrid renewable dis-tributed generator. In Proceedings of the Canadian Conference on Electrical and Computer Engineering, Saskatoon, SK, Canada, 1–4 May 2005; pp. 1477–1481.
- 7. Sahoo, U.; Kumar, R.; Singh, S.; Tripathi, A. Energy, exergy, economic analysis and optimization of polygeneration hybrid solar-biomass system. *Appl. Therm. Eng.* **2018**, *145*, 685–692. [CrossRef]
- Koohi-Kamali, S.; Rahim, N.A. Coordinated control of smart microgrid during and after islanding operation to prevent under frequency load shedding using energy storage system. *Energy Convers. Manag.* 2016, 127, 623–646. [CrossRef]
- Wu, W.; Chen, S.-A.; Chiu, Y.-C. Design and Control of an SOFC/GT Hybrid Power Generation System with Low Carbon Emissions. *Ind. Eng. Chem. Res.* 2016, 55, 1281–1291. [CrossRef]
- Pandiyan, P.; Sitharthan, R.; Saravanan, S.; Prabaharan, N.; Tiwari, M.R.; Chinnadurai, T.; Yuvaraj, T.; Devabalaji, K. A comprehensive review of the prospects for rural electrification using stand-alone and hybrid energy technologies. *Sustain. Energy Technol. Assess.* 2022, 52, 102155. [CrossRef]
- 11. Leal Filho, W.; Mannke, F.; Mohee, R.; Schulte, V.; Surroop, D. Climate-Smart Technologies; Springer: Berlin/Heidelberg, Germany, 2013.
- 12. Bertheau, P.; Blechinger, P. Resilient solar energy island supply to support SDG7 on the Philippines: Techno-economic optimized electrification strategy for small islands. *Util. Policy* **2018**, *54*, 55–77. [CrossRef]
- Das, B.K.; Hoque, N.; Mandal, S.; Pal, T.K.; Raihan, A. A techno-economic feasibility of a stand-alone hybrid power generation for remote area application in Bangladesh. *Energy* 2017, 134, 775–788. [CrossRef]
- 14. Abd El-Sattar, H.; Kamel, S.; Hassan, M.H.; Jurado, F. Optimal sizing of an off-grid hybrid photovoltaic/biomass gasifi-er/battery system using a quantum model of Runge Kutta algorithm. *Energy Convers. Manag.* **2022**, 258, 115539. [CrossRef]
- 15. Jha, S.K.; Bilalovic, J.; Jha, A.; Patel, N.; Zhang, H. Renewable energy: Present research and future scope of Artificial Intelligence. *Renew. Sustain. Energy Rev.* 2017, 77, 297–317. [CrossRef]
- 16. Moria, H.; Yanbu Industrial College. Techno-Economic Optimization of Solar/Wind Turbine System for Remote Mosque in Saudi Arabia Highway: Case Study. *Int. J. Eng. Res.* **2019**, *V8*, 090037. [CrossRef]
- 17. Mamaghani, A.H.; Escandon, S.A.A.; Najafi, B.; Shirazi, A.; Rinaldi, F. Techno-economic feasibility of photovoltaic, wind, diesel and hybrid electrification systems for off-grid rural electrification in Colombia. *Renew. Energy* **2016**, *97*, 293–305. [CrossRef]
- 18. Siddaiah, R.; Saini, R.P. A review on planning, configurations, modeling and optimization techniques of hybrid re-newable energy systems for off grid applications. *Renew. Sustain. Energy Rev.* **2016**, *58*, 376–396. [CrossRef]
- Kazem, H.A.; Chaichan, M.T. Evaluation of grid-connected photovoltaic system in omani harsh weathers CO<sub>2</sub>. AIP Conf. Proc. 2022, 2415, 030006. [CrossRef]
- Kazem, H.A.; Chaichan, M.T.; Al-Waeli, A.H.; Sopian, K. Evaluation of aging and performance of grid-connected pho-tovoltaic system northern Oman: Seven years' experimental study. Sol. Energy 2020, 207, 1247–1258. [CrossRef]
- Wang, G.; Xin, H.; Wu, D.; Ju, P. Data-driven probabilistic small signal stability analysis for grid-connected PV systems. *Int. J. Electr. Power Energy Syst.* 2019, 113, 824–831. [CrossRef]
- Su, S.; Yan, X.; Agbossou, K.; Chahine, R.; Zong, Y. Artificial intelligence for hydrogen-based hybrid renewable energy systems: A review with case study. J. Physics Conf. Ser. 2022, 2208, 012013. [CrossRef]
- Sinha, S.; Chandel, S. Review of recent trends in optimization techniques for solar photovoltaic–wind based hybrid energy systems. *Renew. Sustain. Energy Rev.* 2015, 50, 755–769. [CrossRef]
- 24. Bashar, A.; Smys, S. Integrated renewable energy system for stand-alone operations with optimal load dispatch strategy. *J. Electron. Inform.* **2021**, *3*, 89–98. [CrossRef]
- Ullah, Z.; Elkadeem, M.R.; Kotb, K.M.; Taha, I.B.; Wang, S. Multi-criteria decision-making model for optimal planning of on/off grid hybrid solar, wind, hydro, biomass clean electricity supply. *Renew. Energy* 2021, 179, 885–910. [CrossRef]
- Al-Quraan, A.; Al-Qaisi, M. Modelling, Design and Control of a Standalone Hybrid PV-Wind Micro-Grid System. *Energies* 2021, 14, 4849. [CrossRef]
- Yan, B.; Luh, P.B.; Warner, G.; Zhang, P. Operation and Design Optimization of Microgrids With Renewables. *IEEE Trans. Autom. Sci. Eng.* 2017, 14, 573–585. [CrossRef]
- Pascasio, J.D.A.; Esparcia, E.A.; Castro, M.T.; Ocon, J.D. Comparative assessment of solar photovoltaic-wind hybrid energy systems: A case for Philippine off-grid islands. *Renew. Energy* 2021, 179, 1589–1607. [CrossRef]
- Amer, M.; Namaane, A.; M'Sirdi, N. Optimization of Hybrid Renewable Energy Systems (HRES) Using PSO for Cost Reduction. Energy Procedia 2013, 42, 318–327. [CrossRef]
- Marocco, P.; Ferrero, D.; Lanzini, A.; Santarelli, M. Optimal design of stand-alone solutions based on RES + hydrogen storage feeding off-grid communities. *Energy Convers. Manag.* 2021, 238, 114147. [CrossRef]
- Bohre, A.K.; Sawle, Y.; Acharjee, P. Optimal design and techno-socio-economic analysis of hybrid renewable system for girdconnected system. *Renew. Energy Syst.* 2021, 653–686. [CrossRef]
- 32. Fares, D.; Fathi, M.; Mekhilef, S. Performance evaluation of metaheuristic techniques for optimal sizing of a stand-alone hybrid PV/wind/battery system. *Appl. Energy* **2021**, *305*, 117823. [CrossRef]
- Riaz, M.; Hanif, A.; Hussain, S.J.; Memon, M.I.; Ali, M.U.; Zafar, A. An Optimization-Based Strategy for Solving Optimal Power Flow Problems in a Power System Integrated with Stochastic Solar and Wind Power Energy. *Appl. Sci.* 2021, 11, 6883. [CrossRef]
- 34. Cai, W.; Li, C.; Agbossou, K.; Bénard, P.; Xiao, J. A review of hydrogen-based hybrid renewable energy systems: Simulation and optimization with artificial intelligence. *J. Physics Conf. Ser.* **2022**, 2208, 012012. [CrossRef]

- 35. Frimpong, S.O.; Millham, R.C.; Agbehadji, I.E. A Comprehensive Review of Nature-Inspired Search Techniques Used in Estimating Optimal Configuration Size, Cost, and Reliability of a Mini-grid HRES: A Systemic Review. In *Computational Science and Its Applications—ICCSA 2021*; Springer: Cham, Switzerland, 2021; pp. 492–507. [CrossRef]
- 36. Umar, T. Sustainable energy production from municipal solid waste in Oman. *Proc. Inst. Civ. Eng. Eng. Sustain.* 2022, 175, 3–11.
- Umar, T.; Egbu, C.; Ofori, G.; Honnurvali, M.S.; Saidani, M.; Opoku, A. Challenges towards renewable energy: An exploratory study from the Arabian Gulf region. *Proc. Inst. Civ. Eng. Energy* 2020, 173, 68–80. [CrossRef]
- Erdinc, O.; Uzunoglu, M. Optimum design of hybrid renewable energy systems: Overview of different approaches. *Renew. Sustain. Energy Rev.* 2012, 16, 1412–1425. [CrossRef]
- 39. Upadhyay, S.; Sharma, M. A review on configurations, control and sizing methodologies of hybrid energy systems. *Renew. Sustain. Energy Rev.* 2014, *38*, 47–63. [CrossRef]
- 40. Lin, Y.-H.; Lin, M.-D.; Tsai, K.-T.; Deng, M.-J.; Ishii, H. Multi-objective optimization design of green building envelopes and air conditioning systems for energy conservation and CO2 emission reduction. *Sustain. Cities Soc.* **2020**, *64*, 102555. [CrossRef]
- Al-Othman, A.; Tawalbeh, M.; Martis, R.; Dhou, S.; Orhan, M.; Qasim, M.; Olabi, A.G. Artificial intelligence and numerical models in hybrid renewable energy systems with fuel cells: Ad-vances and prospects. *Energy Convers. Manag.* 2022, 253, 115154. [CrossRef]
- 42. Kaur, H.; Gupta, S.; Dhingra, A. Analysis of hybrid solar biomass power plant for generation of electric power. *Mater. Today: Proc.* **2021**, *48*, 1134–1140. [CrossRef]
- 43. Kumar, R.; Channi, H.K. A PV-Biomass off-grid hybrid renewable energy system (HRES) for rural electrification: Design, optimization and techno-economic-environmental analysis. *J. Clean. Prod.* **2022**, *349*, 131347. [CrossRef]
- 44. Kirim, Y.; Sadikoglu, H.; Melikoglu, M. Technical and economic analysis of biogas and solar photovoltaic (PV) hybrid renewable energy system for dairy cattle barns. *Renew. Energy* **2022**, *188*, 873–889. [CrossRef]
- 45. Tazvinga, H.; Dzobo, O. Feasibility Study of a Solar-Biogas System for Off-Grid Applications. In Proceedings of the 2019 9th International Conference on Power and Energy Systems (ICPES), Perth, WA, Australia, 10–12 December 2019; pp. 1–5.
- 46. Somusekhar, R.K.G. Design of Off-Grid Village With Bio-Solar Hybrid Energy. Int. J. Eng. Res. 2016, V5, 040523. [CrossRef]
- 47. Rahmana, M.M.; Hasana, M.M.; Paateroa, J.V.; Lahdelmaa, R. Hybrid application of biogas and solar resources for ful-filling household energy needs: A potentially viable option in rural areas. *Renew. Energy* **2014**, *68*, 35–45. [CrossRef]
- 48. Ansori, A.; Yunitasari, B.; Soeryanto; Muhaji. Environmentally Friendly Power Generation Technology with Solar PV-Biogas in Rural Areas of Eastern Java. *IOP Conf. Series: Earth Environ. Sci.* **2019**, 239, 012030. [CrossRef]
- Ansari, M.S.; Jalil, M.F.; Bansal, R. A review of optimization techniques for hybrid renewable energy systems. *Int. J. Model. Simul.* 2022, 1–14. [CrossRef]
- 50. Bansal, A.K. Sizing and forecasting techniques in photovoltaic-wind based hybrid renewable energy system: A review. *J. Clean. Prod.* **2022**, *369*, 133376. [CrossRef]
- 51. Kumar, P.; Pal, N.; Sharma, H. Optimization and techno-economic analysis of a solar photo-voltaic/biomass/diesel/battery hybrid off-grid power generation system for rural remote electrification in eastern India. *Energy* **2022**, 247, 123560. [CrossRef]
- 52. Yuan, J.; Xu, J.; Wang, Y. Techno-economic study of a distributed hybrid renewable energy system supplying electrical power and heat for a rural house in China. *IOP Conf. Series: Earth Environ. Sci.* **2018**, *127*, 012001. [CrossRef]
- 53. Nuvvula, R.S.S.; Devaraj, E.; Teegala, S.K. A hybrid multiobjective optimization technique for optimal sizing of BESS-WtE supported multi-MW HRES to overcome ramp rate limitations on thermal stations. *Int. Trans. Electr. Energy Syst.* 2021, *31*, e13241. [CrossRef]
- 54. Babaei, R.; Ting, D.S.-K.; Carriveau, R. Feasibility and optimal sizing analysis of stand-alone hybrid energy systems coupled with various battery technologies: A case study of Pelee Island. *Energy Rep.* **2022**, *8*, 4747–4762. [CrossRef]
- 55. Rezaei, M.; Dampage, U.; Das, B.K.; Nasif, O.; Borowski, P.F.; Mohamed, M.A. Investigating the Impact of Economic Uncertainty on Optimal Sizing of Grid-Independent Hybrid Renewable Energy Systems. *Processes* **2021**, *9*, 1468. [CrossRef]
- 56. Bakht, M.P.; Salam, Z.; Bhatti, A.R.; Sheikh, U.U.; Khan, N.; Anjum, W. Techno-economic modelling of hybrid energy system to overcome the load shedding problem: A case study of Pakistan. *PLoS ONE* **2022**, *17*, e0266660. [CrossRef] [PubMed]
- 57. Ali, F.; Ahmar, M.; Jiang, Y.; AlAhmad, M. A techno-economic assessment of hybrid energy systems in rural Pakistan. *Energy* **2020**, *215*, 119103. [CrossRef]
- 58. Li, G.; Yuan, B.; Ge, M.; Xiao, G.; Li, T.; Wang, J.-Q. Capacity configuration optimization of a hybrid renewable energy system with hydrogen storage. *Int. J. Green Energy* **2022**, *19*, 1583–1599. [CrossRef]
- 59. Román, V.B.; Baños, G.E.; Solís, C.Q.; Flota-Bañuelos, M.; Rivero, M.; Soberanis, M.E. Comparative study on the cost of hybrid energy and energy storage systems in remote rural communities near Yucatan, Mexico. *Appl. Energy* **2022**, *308*, 118334. [CrossRef]
- 60. Das, U.; Mandal, S.; Bhattacharjee, S.; Nandi, C. A review of different configuration of hybrid energy systems with case study analysis. *Int. J. Environ. Sustain. Dev.* **2021**, *21*, 116. [CrossRef]
- Feng, L.; Zhang, X.; Li, X.; Li, B.; Li, Y.; Xu, Y.; Guo, H.; Zhou, X.; Chen, H. Performance analysis of hybrid energy storage integrated with distributed renewable energy. *Energy Rep.* 2022, *8*, 1829–1838. [CrossRef]
- 62. Debnath, D.; Ray, S. Hybrid Energy System for an Academic Institution: A Case Study. In *Renewable Energy Optimization, Planning and Control. Studies in Infrastructure and Control;* Khosla, A., Aggarwal, M., Eds.; Springer: Singapore, 2021; pp. 31–39. [CrossRef]
- 63. Cebotari, S.; Benedek, J. Renewable Energy Project as a Source of Innovation in Rural Communities: Lessons from the Periphery. *Sustainability* **2017**, *9*, 509. [CrossRef]

- 64. Poggi, F.; Firmino, A.; Amado, M. Planning renewable energy in rural areas: Impacts on occupation and land use. *Energy* **2018**, 155, 630–640. [CrossRef]
- 65. Delicado, A.; Figueiredo, E.; Silva, L. Community perceptions of renewable energies in Portugal: Impacts on environment, landscape and local development. *Energy Res. Soc. Sci.* 2016, 13, 84–93. [CrossRef]
- 66. Xu, D.; Zhou, B.; Chan, K.W.; Li, C.; Wu, Q.; Chen, B.; Xia, S. Distributed Multienergy Coordination of Multimicrogrids With Biogas-Solar-Wind Renewables. *IEEE Trans. Ind. Inform.* **2018**, *15*, 3254–3266. [CrossRef]
- 67. Mohammad-Alikhani, A.; Mahmoudi, A.; Khezri, R.; Kahourzade, S. Multiobjective Optimization of System Configuration and Component Capacity in an AC Minigrid Hybrid Power System. *IEEE Trans. Ind. Appl.* **2022**, *58*, 4158–4170. [CrossRef]
- Javeed, I.; Khezri, R.; Mahmoudi, A.; Yazdani, A.; Shafiullah, G. Optimal Sizing of Rooftop PV and Battery Storage for Grid-Connected Houses Considering Flat and Time-of-Use Electricity Rates. *Energies* 2021, 14, 3520. [CrossRef]
- Khezri, R.; Mahmoudi, A.; Haque, M.H. Two-Stage Optimal Sizing of Standalone Hybrid Electricity Systems with Time-of-Use Incentive Demand Response. In Proceedings of the 2020 IEEE Energy Conversion Congress and Exposition (ECCE), Detroit, MI, USA, 11–15 October 2020; pp. 2759–2765. [CrossRef]
- 70. Zhao, S.; Sun, W.; Li, J.; Gong, Y. Dynamic modeling of a proton exchange membrane fuel cell using chaotic binary shark smell optimizer from electrical and thermal viewpoints. *Int. J. Energy Environ. Eng.* **2022**, *13*, 1067–1080. [CrossRef]
- 71. Alramlawi, M.; Li, P. Design Optimization of a Residential PV-Battery Microgrid With a Detailed Battery Lifetime Estimation Model. *IEEE Trans. Ind. Appl.* **2020**, *56*, 2020–2030. [CrossRef]
- 72. Zhang, G.; Wang, W.; Du, J.; Liu, H. A multiobjective optimal operation of a stand-alone microgrid using SAPSO algo-rithm. *J. Electr. Comput. Eng.* **2020**, 2020, 6042105. [CrossRef]
- 73. Khezri, R.; Mahmoudi, A.; Aki, H.; Muyeen, S.M. Optimal Planning of Remote Area Electricity Supply Systems: Com-prehensive Review, Recent Developments and Future Scopes. *Energies* **2021**, *14*, 5900. [CrossRef]
- El-houari, H.; Allouhi, A.; Rehman, S.; Buker, M.S.; Kousksou, T.; Jamil, A.; El Amrani, B. Design, simulation, and economic optimization of an off-grid photovoltaic system for rural electrification. *Energies* 2019, 12, 4735. [CrossRef]
- El-Houari, H.; Allouhi, A.; Rehman, S.; Buker, M.; Kousksou, T.; Jamil, A.; El Amrani, B. Feasibility evaluation of a hybrid renewable power generation system for sustainable electricity supply in a Moroccan remote site. *J. Clean. Prod.* 2020, 277, 123534. [CrossRef]
- 76. Khezri, R.; Mahmoudi, A.; Aki, H. Optimal planning of solar photovoltaic and battery storage systems for grid-connected residential sector: Review, challenges and new perspectives. *Renew. Sustain. Energy Rev.* **2021**, 153, 111763. [CrossRef]
- 77. Khan, A.; Javaid, N. Jaya Learning-Based Optimization for Optimal Sizing of Stand-Alone Photovoltaic, Wind Turbine, and Battery Systems. *Engineering* **2020**, *6*, 812–826. [CrossRef]
- Sadeghian, O.; Shotorbani, A.M.; Mohammadi-Ivatloo, B. Risk-averse scheduling of virtual power plants considering electric vehicles and demand response. *Sched. Oper. Virtual Power Plants* 2022, 54, 227–256. [CrossRef]
- Kumar, A.; Kumar, K.; Kapoor, N.R. Optimization of renewable energy sources using emerging computational tech-niques. In Sustainable Developments by Artificial Intelligence and Machine Learning for Renewable Energies; Elsevier: Amsterdam, The Netherlands, 2022; pp. 187–236.
- 80. Paliwal, P. A Technical Review on Reliability and Economic Assessment Framework of Hybrid Power System with Solar and Wind Based Distributed Generators. *Int. J. Integr. Eng.* **2021**, *13*, 233–252. [CrossRef]
- Sadeghi, D.; Ahmadi, S.E.; Amiri, N.; Marzband, M.; Abusorrah, A.; Rawa, M. Designing, optimizing and comparing dis-tributed generation technologies as a substitute system for reducing life cycle costs, CO<sub>2</sub> emissions, and power losses in residential buildings. *Energy* 2022, 253, 123947. [CrossRef]
- 82. Hassan, A.; Al-Abdeli, Y.M.; Masek, M.; Bass, O. Optimal sizing and energy scheduling of grid-supplemented solar PV systems with battery storage: Sensitivity of reliability and financial constraints. *Energy* **2022**, *238*, 121780. [CrossRef]
- 83. Fotopoulou, M.; Rakopoulos, D.; Stergiopoulos, F.; Voutetakis, S. A Review on the Driving Forces, Challenges, and Ap-plications of AC/DC Hybrid Smart Microgrids. In *Smart Grids Technology and Applications*; John Wiley & Sons: Hoboken, NJ, USA, 2022.
- 84. Monteiro, V.; Martins, J.S.; Fernandes, J.C.A.; Afonso, J.L. Review of a Disruptive Vision of Future Power Grids: A New Path Based on Hybrid AC/DC Grids and Solid-State Transformers. *Sustainability* **2021**, *13*, 9423. [CrossRef]
- Bouaouda, A.; Sayouti, Y. Hybrid Meta-Heuristic Algorithms for Optimal Sizing of Hybrid Renewable Energy System: A Review of the State-of-the-Art. Arch. Comput. Methods Eng. 2022, 29, 4049–4083. [CrossRef]
- 86. Gbadamosi, S.L.; Nwulu, N.I. Optimal Configuration of Hybrid Energy System for Rural Electrification of Community Healthcare Facilities. *Appl. Sci.* 2022, 12, 4262. [CrossRef]
- 87. Seedahmed, M.M.; Ramli, M.A.; Bouchekara, H.R.; Milyani, A.H.; Rawa, M.; Budiman, F.N.; Muktiadji, R.F.; Hassan, S.M.U. Optimal sizing of grid-connected photovoltaic system for a large commercial load in Saudi Arabia. *Alex. Eng. J.* 2021, *61*, 6523–6540. [CrossRef]
- 88. El Boujdaini, L.; Mezrhab, A.; Moussaoui, M.A.; Jurado, F.; Vera, D. Sizing of a stand-alone PV–wind–battery–diesel hybrid energy system and optimal combination using a particle swarm optimization algorithm. *Electr. Eng.* **2022**, *104*, 3339–3359. [CrossRef]
- Sahoo, S.; Swain, S.C.; Chowdary, K.V.; Pradhan, A. Cost and Feasibility Analysis for Designing a PV–Wind Hybrid Renewable Energy System (A Case Study for Campus-3, KIIT University, Bhubaneswar). In *Innovation in Electrical Power Engineering*, *Communication, and Computing Technology*; Springer: Berlin/Heidelberg, Germany, 2022; pp. 243–253.

- 90. Kefif, N.; Melzi, B.; Hashemian, M.; Assad, M.E.H.; Hoseinzadeh, S. Feasibility and optimal operation of micro energy hybrid system (hydro/wind) in the rural valley region. *Int. J. Low-Carbon Technol.* **2021**, *17*, 58–68. [CrossRef]
- Al-Najjar, H.; Pfeifer, C.; Al Afif, R.; El-Khozondar, H.J. Performance Evaluation of a Hybrid Grid-Connected Photovoltaic Biogas-Generator Power System. *Energies* 2022, 15, 3151. [CrossRef]
- Ren, Y.; Yao, X.; Liu, D.; Qiao, R.; Zhang, L.; Zhang, K.; Jin, K.; Li, H.; Ran, Y.; Li, F. Optimal design of hydro-wind-PV multi-energy complementary systems considering smooth power output. *Sustain. Energy Technol. Assess.* 2022, 50, 101832. [CrossRef]
- Pujari, H.K.; Rudramoorthy, M. Optimal design and techno-economic analysis of a hybrid grid-independent renewable energy system for a rural community. *Int. Trans. Electr. Energy Syst.* 2021, 31, e13007. [CrossRef]
- 94. Hassan, R.; Das, B.K.; Hasan, M. Integrated off-grid hybrid renewable energy system optimization based on economic, environmental, and social indicators for sustainable development. *Energy* **2022**, *250*, 123823. [CrossRef]
- Olatomiwa, L.; Mekhilef, S.; Huda, A.S.N.; Sanusi, K. Techno-economic analysis of hybrid PV –diesel–battery and PV–wind– diesel–battery power systems for mobile BTS: The way forward for rural development. *Energy Sci. Eng.* 2015, *3*, 271–285. [CrossRef]
- Rehman, S.; Habib, H.U.R.; Wang, S.; Buker, M.S.; Alhems, L.M.; Al Garni, H.Z. Optimal Design and Model Predictive Control of Standalone HRES: A Real Case Study for Residential Demand Side Management. *IEEE Access* 2020, *8*, 29767–29814. [CrossRef]
- 97. Das, B.K.; Tushar, M.S.H.; Hassan, R. Techno-economic optimisation of stand-alone hybrid renewable energy systems for concurrently meeting electric and heating demand. *Sustain. Cities Soc.* 2021, *68*, 102763. [CrossRef]
- 98. Kumar, S.; Sethuraman, C.; Chandru, G. Design of Optimum Sizing for Hybrid Renewable Energy System using HOMER Pro to Meet the Identical Load Demand at Selected Indian Cities. *Int. J. Grid Distrib. Comput.* **2021**, *14*, 1589–1607.
- 99. Raff, R.; Golub, V.; Knežević, G.; Topić, D. Modeling of the Off-Grid PV-Wind-Battery System Regarding Value of Loss of Load Probability. *Energies* **2022**, *15*, 795. [CrossRef]
- 100. Mubaarak, S.; Zhang, D.; Chen, Y.; Liu, J.; Wang, L.; Yuan, R.; Wu, J.; Zhang, Y.; Li, M. Techno-Economic Analysis of Grid-Connected PV and Fuel Cell Hybrid System Using Different PV Tracking Techniques. *Appl. Sci.* **2020**, *10*, 8515. [CrossRef]
- Zhu, T.; Wills, R.G.; Lot, R.; Kong, X.; Yan, X. Optimal sizing and sensitivity analysis of a battery-supercapacitor energy storage system for electric vehicles. *Energy* 2021, 221, 119851. [CrossRef]
- 102. Jahangir, M.H.; Javanshir, F.; Kargarzadeh, A. Economic analysis and optimal design of hydrogen/diesel backup system to improve energy hubs providing the demands of sport complexes. *Int. J. Hydrogen Energy* **2021**, *46*, 14109–14129. [CrossRef]
- Pandyaswargo, A.H.; Wibowo, A.D.; Onoda, H. Socio-techno-economic assessment to design an appropriate renewable energy system for remote agricultural communities in developing countries. *Sustain. Prod. Consum.* 2022, 31, 492–511. [CrossRef]
- 104. Goel, S.; Sharma, R. Optimal sizing of a biomass–biogas hybrid system for sustainable power supply to a commercial agricultural farm in northern Odisha, India. *Environ. Dev. Sustain.* 2019, 21, 2297–2319. [CrossRef]
- 105. Al-Ghussain, L.; Ahmad, A.D.; Abubaker, A.M.; Mohamed, M.A. An integrated photovoltaic/wind/biomass and hybrid energy storage systems towards 100% renewable energy microgrids in university campuses. *Sustain. Energy Technol. Assess.* 2021, 46, 101273. [CrossRef]
- 106. Rashid, M.U.; Ullah, I.; Mehran, M.; Baharom, M.N.R.; Khan, F. Techno-Economic Analysis of Grid-Connected Hybrid Renewable Energy System for Remote Areas Electrification Using Homer Pro. J. Electr. Eng. Technol. 2022, 17, 981–997. [CrossRef]
- 107. Shah, S.; Mahajan, D.; Varun, R.; Jain, V.; Sawle, Y. Optimal Planning and Design of an Off-Grid Solar, Wind, Biomass, Fuel Cell Hybrid Energy System Using HOMER Pro. In *Recent Advances in Power Systems*; Springer: Berlin/Heidelberg, Germany, 2022; pp. 255–275.
- 108. Pujari, H.K.; Rudramoorthy, M. Optimal design, prefeasibility techno-economic and sensitivity analysis of off-grid hybrid renewable energy system. *Int. J. Sustain. Energy* **2022**, *41*, 1466–1498. [CrossRef]
- 109. Tay, G.; Acakpovi, A.; Adjei, P.; Aggrey, G.K.; Sowah, R.; Kofi, D.; Afonope, M.; Sulley, M. Optimal sizing and techno-economic analysis of a hybrid solar PV/wind/diesel generator system. *IOP Conf. Ser. Earth Environ. Sci.* 2022, 1042, 012014. [CrossRef]
- Hutasuhut, A.A.; Riandra, J.; Irwanto, M. Analysis of hybrid power plant scheduling system die-sel/photovoltaic/microhydro in remote area. J. Phys. Conf. Ser. 2022, 2193, 012024. [CrossRef]
- 111. Mahmud, S.; Kaihan, M.K.; Salehin, S.; Ferdaous, M.T.; Nasim, M. Hybrid renewable energy systems for a remote community in a high mountain plateau. *Int. J. Energy Environ. Eng.* 2022, *13*, 1335–1348. [CrossRef]
- Babatunde, O.; Denwigwe, I.; Oyebode, O.; Ighravwe, D.; Ohiaeri, A.; Babatunde, D. Assessing the use of hybrid renewable energy system with battery storage for power generation in a University in Nigeria. *Environ. Sci. Pollut. Res.* 2021, 29, 4291–4310. [CrossRef] [PubMed]
- Khan, F.A.; Pal, N.; Saeed, S.H. Optimization and sizing of SPV/Wind hybrid renewable energy system: A tech-no-economic and social perspective. *Energy* 2021, 233, 121114. [CrossRef]
- 114. Osaretin, C.A.; Iqbal, T.; Butt, S. Optimal sizing and techno-economic analysis of a renewable power system for a remote oil well. *AIMS Electron. Electr. Eng.* **2020**, *4*, 132–153. [CrossRef]
- 115. Yasin, A.; Alsayed, M. Optimization with excess electricity management of a PV, energy storage and diesel generator hybrid system using HOMER Pro software. *Int. J. Appl. Power Eng. (IJAPE)* **2020**, *9*, 267–283. [CrossRef]
- Anoune, K.; Laknizi, A.; Bouya, M.; Astito, A.; Ben Abdellah, A. Sizing a PV-Wind based hybrid system using deterministic approach. *Energy Convers. Manag.* 2018, 169, 137–148. [CrossRef]
- 117. Fathy, A.; Kaaniche, K.; Alanazi, T.M. Recent Approach Based Social Spider Optimizer for Optimal Sizing of Hybrid PV/Wind/Battery/Diesel Integrated Microgrid in Aljouf Region. *IEEE Access* 2020, *8*, 57630–57645. [CrossRef]

- Moghaddam, M.J.H.; Kalam, A.; Nowdeh, S.A.; Ahmadi, A.; Babanezhad, M.; Saha, S. Optimal sizing and energy management of stand-alone hybrid photovoltaic/wind system based on hydrogen storage considering LOEE and LOLE reliability indices using flower pollination algorithm. *Renew. Energy* 2019, 135, 1412–1434. [CrossRef]
- Fioriti, D.; Giglioli, R.; Poli, D.; Lutzemberger, G.; Vanni, A.; Salza, P. Optimal sizing of a mini-grid in developing countries, taking into account the operation of an electrochemical storage and a fuel tank. In Proceedings of the 2017 6th International Conference on Clean Electrical Power (ICCEP), Santa Margherita Ligure, Italy, 27–29 June 2017; pp. 320–326. [CrossRef]
- 120. Sawle, Y.; Gupta, S.; Bohre, A.K. Optimal sizing of standalone PV/Wind/Biomass hybrid energy system using GA and PSO optimization technique. *Energy Procedia* 2017, 117, 690–698. [CrossRef]
- 121. Yimen, N.; Tchotang, T.; Kanmogne, A.; Idriss, I.A.; Musa, B.; Aliyu, A.; Okonkwo, E.; Abba, S.; Tata, D.; Meva'A, L.; et al. Optimal Sizing and Techno-Economic Analysis of Hybrid Renewable Energy Systems—A Case Study of a Photo-voltaic/Wind/Battery/Diesel System in Fanisau, Northern Nigeria. *Processes* **2020**, *8*, 1381. [CrossRef]
- 122. Emad, D.; El-Hameed, M.A.; Yousef, M.T.; El-Fergany, A.A. Computational Methods for Optimal Planning of Hybrid Renewable Microgrids: A Comprehensive Review and Challenges. *Arch. Comput. Methods Eng.* **2020**, *27*, 1297–1319. [CrossRef]
- 123. Kharrich, M.; Kamel, S.; Abdeen, M.; Mohammed, O.H.; Akherraz, M.; Khurshaid, T.; Rhee, S.-B. Developed Approach Based on Equilibrium Optimizer for Optimal Design of Hybrid PV/Wind/Diesel/Battery Microgrid in Dakhla, Morocco. *IEEE Access* 2021, 9, 13655–13670. [CrossRef]
- Paliwal, P. Techno-Socio-Economic Sizing of Solar–Diesel Generator-Based Autonomous Power System Using Butter-fly-PSO. In Advances in Renewable Energy and Electric Vehicles; Springer: Berlin/Heidelberg, Germany, 2022; pp. 427–437.
- 125. Mohseni, S.; Brent, A.C.; Burmester, D.; Browne, W.N.; Kelly, S. Adding a Computationally-Tractable Probabilistic Dimension to Meta-Heuristic-Based Microgrid Sizing. In Proceedings of the TENCON 2021—2021 IEEE Region 10 Conference (TENCON), Auckland, New Zealand, 7–10 December 2021; pp. 464–469. [CrossRef]
- 126. Das, B.K.; Hassan, R.; Islam, S.; Rezaei, M. Influence of energy management strategies and storage devices on the technoenviro-economic optimization of hybrid energy systems: A case study in Western Australia. J. Energy Storage 2022, 51, 104239. [CrossRef]
- Udeh, G.T.; Michailos, S.; Ingham, D.; Hughes, K.J.; Ma, L.; Pourkashanian, M. A modified rule-based energy management scheme for optimal operation of a hybrid PV-wind-Stirling engine integrated multi-carrier energy system. *Appl. Energy* 2022, 312, 118763. [CrossRef]
- 128. Sharma, R.; Kodamana, H.; Ramteke, M. Multi-objective dynamic optimization of hybrid renewable energy systems. *Chem. Eng. Process. Process. Intensif.* **2022**, *170*, 108663. [CrossRef]
- 129. Alshammari, N.; Asumadu, J. Optimum unit sizing of hybrid renewable energy system utilizing harmony search, Jaya and particle swarm optimization algorithms. *Sustain. Cities Soc.* **2020**, *60*, 102255. [CrossRef]
- 130. Maleki, A.; Askarzadeh, A. Comparative study of artificial intelligence techniques for sizing of a hydrogen-based stand-alone photovoltaic/wind hybrid system. *Int. J. Hydrogen Energy* **2014**, *39*, 9973–9984. [CrossRef]
- 131. Zhang, Y.; Zhou, H.; Xiao, L.; Zhao, G. Research on Economic Optimal Dispatching of Microgrid Cluster Based on Improved Butterfly Optimization Algorithm. *Int. Trans. Electr. Energy Syst.* **2022**, 2022, 1–16. [CrossRef]
- 132. Jahannoosh, M.; Nowdeh, S.A.; Naderipour, A.; Kamyab, H.; Davoudkhani, I.F.; Klemeš, J.J. New hybrid meta-heuristic algorithm for reliable and cost-effective designing of photovoltaic/wind/fuel cell energy system considering load interruption probability. *J. Clean. Prod.* **2021**, *278*, 123406. [CrossRef]
- 133. Maleki, A. Design and optimization of autonomous solar-wind-reverse osmosis desalination systems coupling battery and hydrogen energy storage by an improved bee algorithm. *Desalination* **2018**, 435, 221–234. [CrossRef]
- Khezri, R.; Mahmoudi, A.; Haque, M.H. Optimal WT, PV and BES based Energy Systems for Standalone Households in South Australia. In Proceedings of the 2019 IEEE Energy Conversion Congress and Exposition (ECCE), Baltimore, MD, USA, 29 September–3 October 2019; pp. 3475–3482. [CrossRef]
- 135. Lai, C.S.; McCulloch, M.D. Sizing of Stand-Alone Solar PV and Storage System With Anaerobic Digestion Biogas Power Plants. *IEEE Trans. Ind. Electron.* 2016, 64, 2112–2121. [CrossRef]
- Nyeche, E.; Diemuodeke, E. Modelling and optimisation of a hybrid PV-wind turbine-pumped hydro storage energy system for mini-grid application in coastline communities. J. Clean. Prod. 2019, 250, 119578. [CrossRef]
- 137. Zhang, Y.; Sun, H.; Tan, J.; Li, Z.; Hou, W.; Guo, Y. Capacity configuration optimization of multi-energy system integrating wind turbine/photovoltaic/hydrogen/battery. *Energy* **2022**, 252, 124046. [CrossRef]
- 138. Hossain, M.A.; Ahmed, A.; Tito, S.R.; Ahshan, R.; Sakib, T.H.; Nengroo, S.H. Multi-Objective Hybrid Optimization for Optimal Sizing of a Hybrid Renewable Power System for Home Applications. *Energies* **2023**, *16*, 96. [CrossRef]
- 139. Maleki, A.; Askarzadeh, A. Artificial bee swarm optimization for optimum sizing of a stand-alone PV/WT/FC hybrid system considering LPSP concept. *Sol. Energy* **2014**, *107*, 227–235. [CrossRef]
- Su, H.-Y.; Liu, J.-H.; Chu, C.-C.; Lee, S.-H.; Hong, Y.-Y.; Lin, Y.-J.; Liao, C.-J. Developing an Optimal Scheduling of Taiwan Power System With Highly Penetrated Renewable Energy Resources and Pumped Hydro Storages. *IEEE Trans. Ind. Appl.* 2021, 57, 1973–1986. [CrossRef]
- Khemissi, L.; Khiari, B.; Sellami, A. A novel optimal planning methodology of an autonomous Photovoltaic/Wind/Battery hybrid power system by minimizing economic, energetic and environmental objectives. *Int. J. Green Energy* 2021, *18*, 1064–1080. [CrossRef]

- 142. Hamdy, A.S.M.; Sultan, M. Optimal Sizing of Isolated Hybrid PV/WT/FC System Using Manta Ray Foraging Optimization Algorithm. *Int. Trans. J. Eng.* **2020**, *11*, 11A16H. [CrossRef]
- 143. Memon, S.A.; Patel, R.N. An overview of optimization techniques used for sizing of hybrid renewable energy systems. *Renew. Energy Focus* **2021**, *39*, 1–26. [CrossRef]
- Zereg, H.; Bouzgou, H. Multi-Objective Optimization of Stand-Alone Hybrid Renewable Energy System for Rural Electrification in Algeria. In Artificial Intelligence and Heuristics for Smart Energy Efficiency in Smart Cities; Springer: Cham, Switzerland, 2021; pp. 21–33. [CrossRef]
- 145. Samy, M.; Mosaad, M.I.; Barakat, S. Optimal economic study of hybrid PV-wind-fuel cell system integrated to unreliable electric utility using hybrid search optimization technique. *Int. J. Hydrogen Energy* **2021**, *46*, 11217–11231. [CrossRef]
- 146. Jahannoush, M.; Nowdeh, S.A. Optimal designing and management of a stand-alone hybrid energy system using meta-heuristic improved sine–cosine algorithm for Recreational Center, case study for Iran country. *Appl. Soft Comput.* 2020, 96, 106611. [CrossRef]
- 147. Javed, M.S.; Ma, T.; Jurasz, J.; Ahmed, S.; Mikulik, J. Performance comparison of heuristic algorithms for optimization of hybrid off-grid renewable energy systems. *Energy* **2020**, *210*, 118599. [CrossRef]
- 148. Lorestani, A.; Gharehpetian, G.; Nazari, M.H. Optimal sizing and techno-economic analysis of energy- and cost-efficient standalone multi-carrier microgrid. *Energy* **2019**, *178*, 751–764. [CrossRef]
- 149. Ma, T.; Yang, H.; Lu, L.; Peng, J. Technical feasibility study on a standalone hybrid solar-wind system with pumped hydro storage for a remote island in Hong Kong. *Renew. Energy* **2014**, *69*, 7–15. [CrossRef]
- 150. Diab, A.A.Z.; Sultan, H.M.; Kuznetsov, O.N. Optimal sizing of hybrid solar/wind/hydroelectric pumped storage energy system in Egypt based on different meta-heuristic techniques. *Environ. Sci. Pollut. Res.* **2020**, *27*, 32318–32340. [CrossRef]
- Guneser, M.T.; Elbaz, A.; Seker, C. Hybrid Optimization Methods Application on Sizing and Solving the Economic Dispatch Problems of Hybrid Renewable Power Systems. In *Applications of Nature-Inspired Computing in Renewable Energy Systems*; IGI Global: Hershey, PA, USA, 2022; pp. 136–165. [CrossRef]
- 152. Suman, G.K.; Guerrero, J.M.; Roy, O.P. Optimisation of solar/wind/bio-generator/diesel/battery based microgrids for rural areas: A PSO-GWO approach. *Sustain. Cities Soc.* 2021, 67, 102723. [CrossRef]
- 153. Liu, B.; Zhou, B.; Yang, D.; Li, G.; Cao, J.; Bu, S.; Littler, T. Optimal planning of hybrid renewable energy system considering virtual energy storage of desalination plant based on mixed-integer NSGA-III. *Desalination* **2022**, 521, 115382. [CrossRef]
- 154. Pravin, P.; Luo, Z.; Li, L.; Wang, X. Learning-based scheduling of industrial hybrid renewable energy systems. *Comput. Chem. Eng.* **2022**, *159*, 107665. [CrossRef]
- 155. Hemeida, A.M.; Omer, A.S.; Bahaa-Eldin, A.M.; Alkhalaf, S.; Ahmed, M.; Senjyu, T.; El-Saady, G. Multi-objective multi-verse optimization of renewable energy sources-based micro-grid system: Real case. *Ain Shams Eng. J.* 2021, *13*, 101543. [CrossRef]
- 156. Yahiaoui, A.; Tlemçani, A. Superior performances strategies of different hybrid renewable energy systems configurations with energy storage units. *Wind Eng.* **2022**, *46*, 1471–1486. [CrossRef]
- Sun, H.; Ebadi, A.G.; Toughani, M.; Nowdeh, S.A.; Naderipour, A.; Abdullah, A. Designing framework of hybrid photo-voltaicbiowaste energy system with hydrogen storage considering economic and technical indices using whale optimization algorithm. *Energy* 2022, 238, 121555. [CrossRef]
- 158. Alturki, F.A.; Awwad, E.M. Sizing and cost minimization of standalone hybrid wt/pv/biomass/pump-hydro stor-age-based energy systems. *Energies* 2021, 14, 489. [CrossRef]
- Islam, R.; Akter, H.; Howlader, H.O.R.; Senjyu, T. Optimal Sizing and Techno-Economic Analysis of Grid-Independent Hybrid Energy System for Sustained Rural Electrification in Developing Countries: A Case Study in Bangladesh. *Energies* 2022, 15, 6381. [CrossRef]
- Naeem, A.; Hassan, N.U. Renewable Energy Intermittency Mitigation in Microgrids: State-of-the-Art and Future Prospects. In Proceedings of the 2020 4th International Conference on Green Energy and Applications (ICGEA), Singapore, 7–9 March 2020; pp. 158–164. [CrossRef]
- Kaldellis, J.; Zafirakis, D. Optimum energy storage techniques for the improvement of renewable energy sources-based electricity generation economic efficiency. *Energy* 2007, 32, 2295–2305. [CrossRef]
- Hadjipaschalis, I.; Poullikkas, A.; Efthimiou, V. Overview of current and future energy storage technologies for electric power applications. *Renew. Sustain. Energy Rev.* 2009, 13, 1513–1522. [CrossRef]
- 163. Al-Ghussain, L.; Samu, R.; Taylan, O.; Fahrioglu, M. Sizing renewable energy systems with energy storage systems in microgrids for maximum cost-efficient utilization of renewable energy resources. *Sustain. Cities Soc.* **2020**, *55*, 102059. [CrossRef]
- Sánchez, A.; Zhang, Q.; Martín, M.; Vega, P. Towards a new renewable power system using energy storage: An economic and social analysis. *Energy Convers. Manag.* 2022, 252, 115056. [CrossRef]
- 165. Chand, A.A.; Prasad, K.A.; Mamun, K.A.; Sharma, K.R.; Chand, K.K. Adoption of Grid-Tie Solar System at Residential Scale. *Clean Technol.* **2019**, *1*, 224–231. [CrossRef]
- Fossati, J.P.; Galarza, A.; Martín-Villate, A.; Fontán, L. A method for optimal sizing energy storage systems for microgrids. *Renew.* Energy 2015, 77, 539–549. [CrossRef]
- 167. Rosenberg, M.; French, T.; Reynolds, M.; While, L. Finding an optimised infrastructure for electricity distribution networks in rural areas—A comparison of different approaches. *Swarm Evol. Comput.* **2022**, *68*, 101018. [CrossRef]

- 168. Kumar, N.; Chopra, S.; Chand, A.; Elavarasan, R.; Shafiullah, G. Hybrid Renewable Energy Microgrid for a Residential Community: A Techno-Economic and Environmental Perspective in the Context of the SDG7. *Sustainability* 2020, 12, 3944. [CrossRef]
- Hermann, D.T.; Donatien, N.; Armel, T.K.F.; René, T. Techno-economic and environmental feasibility study with demand-side management of photovoltaic/wind/hydroelectricity/battery/diesel: A case study in Sub-Saharan Africa. *Energy Convers. Manag.* 2022, 258, 115494. [CrossRef]
- Kaabeche, A.; Diaf, S.; Ibtiouen, R. Firefly-inspired algorithm for optimal sizing of renewable hybrid system considering reliability criteria. Sol. Energy 2017, 155, 727–738. [CrossRef]
- Das, M.; Singh, M.A.K.; Biswas, A. Techno-economic optimization of an off-grid hybrid renewable energy system using metaheuristic optimization approaches—Case of a radio transmitter station in India. *Energy Convers. Manag.* 2019, 185, 339–352.
   [CrossRef]
- 172. Maleki, A.; Pourfayaz, F.; Ahmadi, M.H. Design of a cost-effective wind/photovoltaic/hydrogen energy system for supplying a desalination unit by a heuristic approach. *Sol. Energy* **2016**, *139*, 666–675. [CrossRef]
- 173. Ridha, H.M.; Gomes, C.; Hazim, H.; Ahmadipour, M. Sizing and implementing off-grid stand-alone photovoltaic/battery systems based on multi-objective optimization and techno-economic (MADE) analysis. *Energy* **2020**, 207, 118163. [CrossRef]
- 174. Kaur, R.; Krishnasamy, V.; Kandasamy, N.K.; Kumar, S. Discrete Multiobjective Grey Wolf Algorithm Based Optimal Sizing and Sensitivity Analysis of PV-Wind-Battery System for Rural Telecom Towers. *IEEE Syst. J.* 2019, 14, 729–737. [CrossRef]
- Guezgouz, M.; Jurasz, J.; Bekkouche, B.; Ma, T.; Javed, M.S.; Kies, A. Optimal hybrid pumped hydro-battery storage scheme for off-grid renewable energy systems. *Energy Convers. Manag.* 2019, 199, 112046. [CrossRef]
- 176. Javed, M.S.; Ma, T.; Jurasz, J.; Amin, M.Y. Solar and wind power generation systems with pumped hydro storage: Review and future perspectives. *Renew. Energy* **2019**, *148*, 176–192. [CrossRef]
- Nazari, M.E.; Ardehali, M.M.; Jafari, S. Pumped-storage unit commitment with considerations for energy demand, economics, and environmental constraints. *Energy* 2010, 35, 4092–4101. [CrossRef]
- 178. Zhang, Z.; Feng, L.; Liu, H.; Wang, L.; Wang, S.; Tang, Z. Mo<sup>6+</sup>–P<sup>5+</sup> co-doped Li<sub>2</sub>ZnTi<sub>3</sub>O<sub>8</sub> anode for Li-storage in a wide temperature range and applications in LiNi<sub>0.5</sub>Mn<sub>1.5</sub>O<sub>4</sub>/Li<sub>2</sub>ZnTi<sub>3</sub>O<sub>8</sub> full cells. *Inorg. Chem. Front.* **2021**, *9*, 35–43. [CrossRef]
- 179. Chen, Y. Research on collaborative innovation of key common technologies in new energy vehicle industry based on digital twin technology. *Energy Rep.* 2022, *8*, 15399–15407. [CrossRef]
- 180. Liu, L.; Tang, Y.; Liu, D. Investigation of future low-carbon and zero-carbon fuels for marine engines from the view of thermal efficiency. *Energy Rep.* **2022**, *8*, 6150–6160. [CrossRef]
- 181. Liu, L.; Wu, Y.; Wang, Y. Numerical investigation on the combustion and emission characteristics of ammonia in a low-speed two-stroke marine engine. *Fuel* **2021**, *314*, 122727. [CrossRef]
- 182. Ge, L.; Li, Y.; Li, Y.; Yan, J.; Sun, Y. Smart Distribution Network Situation Awareness for High-Quality Operation and Maintenance: A Brief Review. *Energies* 2022, 15, 828. [CrossRef]
- Li, X.; Wang, H.; Yang, C. Driving mechanism of digital economy based on regulation algorithm for development of low-carbon industries. *Sustain. Energy Technol. Assess.* 2022, 55, 102909. [CrossRef]
- 184. Lu, S.; Guo, J.; Liu, S.; Yang, B.; Liu, M.; Yin, L.; Zheng, W. An Improved Algorithm of Drift Compensation for Olfactory Sensors. *Appl. Sci.* **2022**, *12*, 9529. [CrossRef]
- 185. Dang, W.; Guo, J.; Liu, M.; Liu, S.; Yang, B.; Yin, L.; Zheng, W. A Semi-Supervised Extreme Learning Machine Algorithm Based on the New Weighted Kernel for Machine Smell. *Appl. Sci.* 2022, *12*, 9213. [CrossRef]
- Lu, S.; Yin, Z.; Liao, S.; Yang, B.; Liu, S.; Liu, M.; Yin, L.; Zheng, W. An asymmetric encoder–decoder model for Zn-ion battery lifetime prediction. *Energy Rep.* 2022, *8*, 33–50. [CrossRef]
- Das, P.; Das, B.K.; Mustafi, N.N.; Sakir, T. A review on pump-hydro storage for renewable and hybrid energy systems applications. Energy Storage 2020, 3, e223. [CrossRef]
- Argyrou, M.C.; Christodoulides, P.; Kalogirou, S.A. Energy storage for electricity generation and related processes: Technologies appraisal and grid scale applications. *Renew. Sustain. Energy Rev.* 2018, 94, 804–821. [CrossRef]
- Nordling, A.; Englund, R.; Hembjer, A.; Mannberg, A. Energy Storage-Electricity Storage Technologies. *IVA'S Electr. Crossroads Proj.* 2016. Available online: https://www.iva.se/globalassets/rapporter/vagval-el/201604-iva-vagvalel-ellagring-rapport-english-e-ny.pdf (accessed on 18 November 2022).
- Medina, P.; Bizuayehu, A.W.; Catalão, J.P.; Rodrigues, E.M.; Contreras, J. Electrical energy storage systems: Technologies' state-ofthe-art, techno-economic benefits and applications analysis. In Proceedings of the 2014 47th Hawaii International Conference on System Sciences, Waikoloa, HI, USA, 6–9 January 2014; pp. 2295–2304.
- Castillo, A.; Gayme, D.F. Grid-scale energy storage applications in renewable energy integration: A survey. *Energy Convers. Manag.* 2014, 87, 885–894. [CrossRef]
- Gallo, A.B.; Simõões-Moreira, J.R.; Costa, H.K.M.; Santos, M.M.; Moutinho dos Santos, E.M. Energy storage in the energy transition context: A technology review. *Renew. Sustain. Energy Rev.* 2016, 65, 800–822. [CrossRef]
- 193. Zhao, P.; Wang, J.; Dai, Y. Capacity allocation of a hybrid energy storage system for power system peak shaving at high wind power penetration level. *Renew. Energy* **2015**, *75*, 541–549. [CrossRef]
- 194. Nguyen, T.-T.; Martin, V.; Malmquist, A.; Silva, C.A. A review on technology maturity of small scale energy storage technologies. *Renew. Energy Environ. Sustain.* 2017, 2, 36. [CrossRef]

- 195. Akinyele, D.O.; Rayudu, R.K. Review of energy storage technologies for sustainable power networks. *Sustain. Energy Technol. Assess.* **2014**, *8*, 74–91. [CrossRef]
- 196. Kousksou, T.; Bruel, P.; Jamil, A.; El Rhafiki, T.; Zeraouli, Y. Energy storage: Applications and challenges. *Sol. Energy Mater. Sol. Cells* **2013**, 120, 59–80. [CrossRef]
- Slaughter, A. Electricity Storage Technologies, Impacts, and Prospects. Deloitte Center for Energy Solutions 2015. Available online: https://www2.deloitte.com/content/dam/Deloitte/us/Documents/energy-resources/us-er-electric-storage-paper.pdf (accessed on 20 November 2022).
- 198. Aneke, M.; Wang, M. Energy storage technologies and real life applications—A state of the art review. *Appl. Energy* 2016, 179, 350–377. [CrossRef]
- 199. Du, P.; Lu, N. Energy Storage for Smart Grids: Planning and Operation for Renewable and Variable Energy Resources (VERs); Academic Press: Cambridge, MA, USA, 2014.
- 200. Hallasmaa, T. Implementation of an Energy Storage for Waste-to-Energy Plant. 2022. Available online: https://urn.fi/URN:NBN: fi:amk-202205169532 (accessed on 25 November 2022).
- 201. Bahramian, P. Integration of Wind Power into an Electricity System Using Pumped Storage: Economic Challenges and Stakeholder Impacts; Queen's Economics Department Working Paper; Queen's University, Department of Economics: Kingston, ON, Canada, 2022.

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.